The international platform of Ludwig-Maximilians University's Center for Economic Studies and the Ifo Institute





Going Deep: The Trade and Welfare Effects of TTIP

Rahel Aichele Gabriel Felbermayr Inga Heiland

CESIFO WORKING PAPER NO. 5150 CATEGORY 8: TRADE POLICY DECEMBER 2014

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com • from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

CESifo Center for Economic Studies & Ifo Institute

Going Deep: The Trade and Welfare Effects of TTIP

Abstract

Since July 2013, the EU and the US have been negotiating a preferential trade agreement (PTA), the Transatlantic Trade and Investment Partnership (TTIP). We use a multi-country, multi-industry Ricardian trade model with national and international input-output linkages to quantify its potential economic consequences. We structurally estimate the sectoral trade flow elasticities of trade costs and of existing PTAs. We simulate the trade, value added, and welfare effects of the TTIP, assuming that the agreement would eliminate all transatlantic tariffs and reduce non-tariff barriers as other deep PTAs have. The long-run level of real per capita income would change by 2.12% in the EU, by 2.68% in the US, and by -0.03% in the rest of the world relative to the status quo. However, there is substantial heterogeneity across the 134 geographical entities that we investigate. Gross value of EU-US trade could triple, but its value added trade than in gross trade. This signals a deepening of the transatlantic value chain.

JEL-Code: F130, F140, F170.

Keywords: structural gravity, preferential trade agreements, TTIP.

Rahel Aichele Ifo Institute – Leibniz Institute for Economic Research at the University of Munich Poschingerstrasse 5 Germany – 81679 Munich aichele@ifo.de

Gabriel Felbermayr Ifo Institute – Leibniz Institute for Economic Research at the University of Munich Poschingerstrasse 5 Germany – 81679 Munich felbermayr@ifo.de Inga Heiland Ifo Institute – Leibniz Institute for Economic Research at the University of Munich Poschingerstrasse 5 Germany – 81679 Munich heiland@ifo.de

We thank Lorenzo Caliendo, Peter Egger, Marc-Andreas Muendler, Mario Larch, and seminar participants in Lisbon, Munich, Rome, Venice, and Vienna for comments and suggestions. This paper provides the technical details for a report that the authors have prepared for the Bertelsmann foundation.

1 Motivation

In July 2013, the EU and the US have begun negotiations on a Transatlantic Trade and Investment Partnership (TTIP). According to the High-Level Working Group (HLWG) on Jobs and Growth, set up by the so called Transatlantic Economic Council (TEC), the ambition is to eliminate all tariffs and to create "... a comprehensive, ambitious agreement that addresses a broad range of bilateral trade and investment issues, including regulatory issues, and contributes to the development of global rules" that "goes beyond what the United States and the EU have achieved in previous trade agreements." In this paper, we attempt a quantification of the potential effects of this endeavor.

TTIP is the first big trade agreement that tries to fill the "gap between 21st century trade and the 20th century trade rules" (Baldwin, 2011) that the relative stasis of the World Trade Organization (WTO) has left developed countries in. Our analysis captures the reality of international production networks and trade in intermediate inputs. It focuses on non-tariff barriers besides tariffs, and it explores scenarios in which the systemic importance of the TTIP also leads to trade cost reductions elsewhere, either through what Francois et al. (2013) have called spillovers, or through the completion of the Doha Round.

And it does so by extending the recent quantitative trade models by Eaton and Kortum (2002) and Caliendo and Parro (2014) to PTAs. Our framework covers 32 industries from the services, manufacturing and agriculture sectors for 134 countries or regions. It incorporates tariffs as well as non-tariff measures (NTMs). By allowing intra- and international trade of intermediate inputs into this stochastic Ricardian model, it allows to model international production networks and allows to differentiate the value added content from the gross value of bilateral trade flows. In contrast to the conventional CGE trade models, the key parameters – the Frechet parameter governing the distribution of productivities within sectors, or the coefficients of the trade cost function – are estimated using structural relationships strictly implied by the theoretical setup.

Using data on sectoral trade flows and input-output linkages from the Global Trade Analysis Project (GTAP) and applying instrumental variables (IV) techniques to obtain unbiased parameter estimates, the central assumption of the analysis is that the TTIP between the EU and the USA will reduce trade costs by as much as other already existing deep trade agreements have. The key results are that the TTIP will yield a long-run increase in the level of real per capita income of 2.12% and 2.68% in the EU and the US, respectively. It would only marginally lower average real income in the rest of the world, leaving the world as a whole better off by about 1.32%. If it were combined with the full elimination of tariffs within the group of WTO countries, almost all countries in the world would win. Similarly, if trade cost reductions between the EU and the US also improve the access of third countries to the TTIP partners or to each other, all could be made better off.

We find that the TTIP would result in a significant amount of trade creation between the insiders. For example, trade between Germany and the US, as measured at the customs, could go up by more than 200%. At the same time, trade with the other EU countries would fall by between 5 and 10%, reflecting trade diversion by preference erosion. Similarly, trade with most third countries or regions would go down. However, imports from suppliers of raw materials or intermediates can go up, reflecting the effect of higher income and of increased industrial production in TTIP partners. Also, trade diversion can be attenuated by imported competitiveness: when TTIP partners supply intermediates at lower prices to third countries, changes in relative prices of final goods are dampened. This latter effect, plus the restructuring of production chains explains the interesting finding that the value added content of trade flows goes up, sometimes substantially, in many trade relationships.

The introduction of a TTIP would alter the composition of aggregate value added. It would lead to a slight reindustrialization in the TTIP partners, reflecting the fact that the reduction in trade barriers is larger in manufacturing than in services. Moreover, the model does not predict that the sectoral impact simply follows the structure of comparative advantage as measured by the Balassa-Samuelson index. Again, this follows from the fact that changes in the competitiveness of sectors is very much driven by changes in the prices of intermediate goods, both domestic and imported.

The paper is related to three important strands of literature. First, it builds on recent work on quantitative trade models. Costinot and Rodríguez-Clare (2014) provide an excellent survey. The central element in these models is the gravity equation, a parsimonious relationship which allows to estimate parameters with the help of relatively simple econometrics but which requires strong functional assumptions. In our case, the Frechet distribution and CES demand systems. The great advantage of the gravity equation is its excellent empirical fit. However, this does not imply that out-of-sample its fit will be perfect as well. Nonetheless, the new "quantitative trade theory" offers important advantages over the more conventional large-scale CGE approach. First, the parsimony allows going relatively far with analytical descriptions. This feature reduces (but does not undo) the black box nature of large general equilibrium models. Second, the approach allows a tight link between theoretical structure and parameter estimation which allows a neater calibration. Finally, by exploiting what we know from existing deep agreements, one does not require bottom-up estimates of NTMs and one can let the data define the TTIP scenario.

Second, our work builds on earlier quantitative evaluations of the TTIP. In a study for the European Commission, Francois et al. (2013) employ a large scale CGE framework based on the well-known GTAP model (Hertel, ed, 1997) and extended to include features from the Francois et al. (2005) model. While their work is at the frontier of classical CGE modeling, it does not utilize the breakthroughs described in Costinot and Rodríguez-Clare (2014). It requires bottom-up estimates of NTMs which are, however, only available for a small set of bilateral trade links, and it defines the scenario on the basis of expert input rather than data. Egger et al. (2014) use the same model, but they rely on a top-down, gravity-based approach to NTMs. However, they do not derive the gravity equation from the model and they seem to use ad hoc calibration to parameterize other model parameters (such as trade elasticities). Moreover, these studies work with regional aggregates.

Felbermayr et al. (2013) and Felbermayr et al. (2014) apply the model and econometric approach of Egger et al. (2011) to simulate the effects of a possible TTIP. The model is a single-sector framework based on the Krugman (1980) model augmented with an extensive margin to capture the prevalence of zero-trade flows. The approach features a tight link between gravity estimation and model structure. However, it does not feature sectoral detail nor does it allow addressing production networks. The work by Anderson et al. (2014) sticks to the single-sector setup but endogenizes the capital stock in a fully structural quantitative trade model. These models have the advantage of great tractability and they can be understood as reduced-form approaches to more complicated setups.

Finally, the present paper relates to a large empirical literature on the determinants and effects of PTAs. Much of the earlier work, as surveyed by Cipollina and Salvatici (2010), is based on reduced form equations and does not properly deal with the potential endogeneity of trade agreements. More recent empirical studies provide a tight link between theoretical model and estimation (see Head and Mayer, 2014), and devote much attention to obtain the causal effects of PTAs on trade flows (see Egger et al., 2011, and the discussion of literature therein). The critical step is to find exogenous drivers of PTA formation. Controlling for tariffs, the estimated treatment effect of PTAs can be used to quantify how PTAs have reduced the costs of non-tariff barriers to trade. Interestingly, the literature suggests that OLS estimates tend to underestimate the true effects of PTAs and that their effect on NTMs must be quite substantial. In our work we provide instrumental variables estimates for 32 sectors (including services) and we distinguish between two types of PTA depth, borrowing a classification provided by Dür et al. (2014).

The remainder of this paper is structured as follows. Section 2 provides a quick overview of the theoretical model. Section 3 discusses the data and the identification of parameters. Section 4 provides the results of the simulation of counterfactual scenarios pertaining to the TTIP. Finally, Section 5 summarizes and concludes.

2 Methodology

We briefly summarize the Eaton and Kortum (2002)-type multi-sector, input-output gravity model developed by Caliendo and Parro (2014) used in our simulations. Their counterfactual analysis deals with the elimination of tariffs between the NAFTA countries. The TTIP will reduce NTMs along with tariffs. It will also provide a deep trade liberalization and go beyond the trade liberalizing effect of many existing PTAs. So to better capture these tariff and NTM trade cost reductions, we introduce PTAs of different depth into the Caliendo and Parro (2014) framework. We characterize the equilibrium changes after a trade policy shock to pave the path for our counterfactual analysis.

Compared to one sector models or models without input-output linkages, the model chosen here features additional welfare channels—an intermediates goods and sector linkages channel (see the discussion in Caliendo and Parro, 2014). Global value chains are increasingly important. The model helps to capture the additional effects.

2.1 The Gravity Model

In n = 1, ..., N countries, the utility function of the representative household is described by a Cobb-Douglas function over j = 1, ..., J sectoral composite goods. α_n^j denotes a sector's expenditure share. The household receives labor income I_n and lump-sum tariff rebates.

Each sector j comprises a continuum of varieties. Labor and the composite goods of each sector k = 1, ..., J are the inputs in j's production process. Let $\beta_n^j \in [0, 1]$ denote the cost share of labor and $\gamma_n^{k,j} \in [0, 1]$ the share of sector k in sector j's intermediate costs, with $\sum_{k=1}^{J} \gamma_n^{k,j} = 1$. Then the production function for a variety ω^j is given by

$$q_n^j(\omega^j) = x_n^j(\omega^j)^{-\theta^j} \left[l_n^j(\omega^j) \right]^{\beta_n^j} \left[\prod_{k=1}^J m_n^{k,j}(\omega^j)^{\gamma_n^{k,j}} \right]^{(1-\beta_n^j)}, \tag{1}$$

where $x_n^j(\omega^j)$ denotes the inverse efficiency of variety producer ω^j . The dispersion of efficiencies across varieties is given by $\theta^j \in (0, 1)$. The higher θ^j , the greater the productivity dispersion in sector j. All varieties ω^j are aggregated with a Dixit-Stiglitz CES technology into sector j's composite good. It can be used for production or consumption purposes.

A sector's varieties are internationally traded. Producers search across all countries for the lowest-cost supplier of a variety. When importing good j from country i to country n, the importer has to pay the unit costs c_i^j times the trade costs κ_{in}^j which consist of two parts: ad-valorem tariffs $\tau_{in}^j \geq 0$ and iceberg trade costs $d_{in}^j \geq 1$, with $d_{nn}^j = 1$. Following other gravity applications, we can model iceberg trade costs as a function of bilateral distance, PTAs and other observable trade cost proxies such as sharing a common border, a common language or a common colonial history. So $d_{in}^j = D_{in} \rho^j e^{\delta i \mathbf{Z}_{in}}$, where D_{in} is bilateral distance, and \mathbf{Z}_{in} is a vector collecting dichotomous trade cost proxies. More explicitly, we allow for two types of PTAs: shallow and deep treaties. The respective dummies are denoted by PTA_{shallow} and PTA_{deep} . Since tariffs are an explicit part of trade costs, the PTA dummies capture trade cost reductions that go beyond the reduction or elimination of tariffs. Thus, the PTA dummies capture the trade-enhancing effect of reducing non-tariff trade barriers like sanitary and phytosanitary trade barriers and other technical barriers to trade like regulatory standards or labeling requirements.

The model gives rise to a gravity equation. Country *n*'s expenditure share π_{in}^j for source country *i*'s goods in sector *j* depends on *i*'s price relative to the price index. It can be written as

$$\pi_{in}^{j} = \frac{\lambda_{i}^{j} \left[c_{i}^{j} \kappa_{in}^{j} \right]^{\frac{-1}{\theta j}}}{\sum_{i=1}^{N} \lambda_{i}^{j} \left[c_{i}^{j} \kappa_{in}^{j} \right]^{\frac{-1}{\theta j}}}.$$
(2)

This trade share can be interpreted as the probability that, for country n, the lowest cost supplier of a variety in sector j is trade partner i. The model is closed with goods market clearing and an income-equals-expenditure condition for each country n.

Due to intermediates goods trade, there is double-counting of value added in trade statistics. Koopman et al. (2014) show how to decompose countries' trade flows into various value added and double-counted parts.⁵ Aichele and Heiland (2014) propose to study production networks by looking at the value added flows that are processed into final goods by one region and then absorbed in another one. Following Johnson and Noguera (2012), they derive counterfactual value added trade flows after a trade policy shock in the Caliendo and Parro (2014) model setup. We apply the same methodology and contrast the effects of the TTIP on trade flows, value added trade flows and production networks.

2.2 Comparative Statics in General Equilibrium

We are interested in the trade and welfare effects of the TTIP. In this section, we describe how the model reacts to a trade policy shock. Let $\hat{x} \equiv x'/x$ be the relative change in a variable from its initial level x to the counterfactual level x'. The formation of a PTA implies changes in the tariff schedule and the reduction of NTMs. So the trade cost changes are given by $\hat{\kappa}_{in}^j = \hat{\tau}_{in}^j e^{\delta_{\text{shallow}}^j (PTA'_{\text{shallow},in} - PTA_{\text{shallow},in}) + \delta_{\text{deep}}^j (PTA'_{\text{deep},in} - PTA_{\text{deep},in})}$. Since all trade flows between liberalizing countries benefit from the tariff and NTM cost reductions, the approach implicitly assumes that either rules of origin do not matter or the local content is sufficiently high.

As suggested by Dekle et al. (2008) one can solve for equilibrium changes:⁶

 $^{{}^{5}}$ Following Koopman et al. (2014), Wang et al. (2013) develop a methodology to decompose trade flows at the bilateral and sectoral level.

⁶When solving for the new equilibrium in changes instead of in levels, the set of parameters that have to be estimated is reduced. Information on price levels, iceberg trade costs, or productivity levels are not required.

$$\hat{c}_n^j = \hat{w}_n^{\beta_n^j} \left(\prod_{k=1}^J \left[\hat{p}_n^k \right]^{\gamma_n^{k,j}} \right)^{1-\beta_n^j}, \tag{3}$$

$$\hat{p}_n^j = \left(\sum_{i=1}^N \pi_{in}^j \left[\hat{\kappa}_{in}^j \hat{c}_i^j\right]^{-1/\theta^j}\right)^{-\theta^j}, \qquad (4)$$

$$\hat{\pi}_{in}^{j} = \left(\frac{\hat{c}_{i}^{j}}{\hat{p}_{n}^{j}}\hat{\kappa}_{in}^{j}\right)^{-1/\delta'}, \qquad (5)$$

$$X_n^{j'} = \sum_{k=1}^J \gamma_n^{j,k} (1 - \beta_n^k) \left(\sum_{i=1}^N \frac{\pi_{ni}^{k'}}{1 + \tau_{ni}^{k'}} X_i^{k'} \right) + \alpha_n^j I_n', \tag{6}$$

$$\sum_{j=1}^{J} F_n^{j'} X_n^{j'} + S_n = \sum_{j=1}^{J} \sum_{i=1}^{N} \frac{\pi_{ni}^{j'}}{1 + \tau_{ni}^{j'}} X_i^{j'},$$
(7)

where \hat{w}_n denotes the wage change, X_n^j denotes the sectoral expenditure level, $F_n^j \equiv \sum_{i=1}^N \frac{\pi_{in}^j}{(1+\tau_{in}^j)}$, $I'_n = \hat{w}_n w_n L_n + \sum_{j=1}^J X_n^{j'}(1-F_n^{j'}) - S_n$, L_n is country *n*'s labor force⁷, and S_n is the trade surplus. Equation (3) shows how unit costs react to input price changes, i.e. to wage and intermediate price changes. Trade cost changes affect the sectoral price index p_n^j directly, and also indirectly by affecting unit costs (see Equation (4)). Changes in trade shares result from these trade cost, unit cost and price changes. The strength of the reaction is governed by the productivity dispersion θ^j . A high θ^j implies bigger trade changes. Equation (6) ensures goods market clearing in the new equilibrium and Equation (7) corresponds to the counterfactual income-equals-expenditure or balanced trade condition. The change in real income which is given by

$$\widehat{W}_n = \frac{\widehat{I}_n}{\prod_{j=1}^J (\widehat{p}_n^j)^{\alpha_n^j}}$$
(8)

serves as a statistic to assess welfare changes.

Caliendo and Parro (2014) extend the single-sector solution algorithm proposed by

⁷Labor can move freely between sectors. However, it cannot cross international borders.

Alvarez and Lucas (2007) to solve the system of equations given by (3)-(7). The algorithm starts with an initial guess about a vector of wage changes. With (3) and (4), it then computes price and trade share changes and the new expenditure levels based on those wage changes, evaluates the trade balance condition (7), and then updates the wage change based on the error in the trade balance.

3 Data and Parameter Identification

To simulate the effects of TTIP in general equilibrium, we need to identify the model parameters α , β , γ , θ , and δ_{shallow} and δ_{deep} , and collect data on bilateral trade shares π , tariff levels τ , countries' total value added $w_n L_n$, and trade surpluses S_n . The expenditure shares α and the cost shares β and γ are obtained from input-output tables. θ , δ_{shallow} and δ_{deep} are estimated based on the gravity equation following from the model.

3.1 Data Sources

Our main data source is the Global Trade Analysis Project (GTAP) 8.1 database, which provides sectoral production values, sectoral value added information and bilateral final and intermediate goods trade in producer and consumer prices, including for service sectors. Given this information, we can construct bilateral tariffs, bilateral input-output tables and expenditure levels. The GTAP database was chosen for its rich country detail.⁸ It contains data for 114 countries and 20 aggregate regions (e.g. "Rest of Southeast Asia"). These 134 countries and regions represent the world economy in the year 2007. GTAP distinguishes 58 sectors. We further aggregate the sectors into 32 industries to keep the simulations tractable. The aggregation pertains mostly to agricultural, resource and

⁸The World Input Output Database (WIOD) constitutes an alternative data source. It provides the same information for a sample of 40 countries and the rest of the world for the years 1995-2011. Since we are interested in trade creation and trade diversion and third country effects, we want to have as much country detail as possible.



Figure 1: Status quo of depth of trade integration

Note: The figure plots the index of depth of PTAs as classified by Dür et al. (2014) for country pairs with a PTA in 2007. The depth index counts the number of provisions and ranges from 0-7. The different provisions are: partial scope agreement, substantive provisions on services, investments, standards, public procurement, competition and intellectual property rights.

processed food sectors but we keep the sectoral detail in the manufacturing and services industries.⁹

A key element of our simulations is to distinguish shallow from deep PTAs. The depth of existing PTAs is constructed from the Design of Trade Agreements (DESTA) database (see Dür et al., 2014). The database offers an index of depth which counts the numbers of provisions (partial scope agreement, substantive provisions on services, investments, standards, public procurement, competition and intellectual property rights) the respective PTA covers. The index ranges from 0 to 7, where 0 indicates a partial scope agreement and 7 is the deepest level of integration. We recode this index of depth to obtain two classes of PTAs: shallow and deep agreements. The dummy indicating a shallow PTA switches to one if the depth index lies between 0 and 7. The dummy for

⁹An overview of the sectoral breakdown is provided in Table A1 in the online appendix.

a deep PTA takes the value of one if it index is between 4 and 7.¹⁰ Figure 1 shows the distribution of the depth of existing PTAs for the year 2007. About 10% of the PTAs (i.e., 836 bilateral relations out of the 7997 with a PTA) are classified as deep according to our definition; examples include NAFTA, the EU or USA-Korea. The Andean Community, MERCOSUR or ASEAN are examples for shallow agreements.

3.2 Identification of Trade Cost Parameters

The trade cost parameters θ and δ can be identified from the gravity equation. Take the trade share equation (2), plug in the functional form for trade costs and multiply by the total expenditure X_n^j , to obtain the following log-linearized estimable gravity equation for each sector j:

$$\ln(\pi_{in}^{j}X_{n}^{j}) = -\frac{1}{\theta^{j}}\ln\tau_{in}^{j} - \frac{\rho^{j}}{\theta^{j}}\ln D_{in} - \frac{\boldsymbol{\delta_{d}}^{j}}{\theta^{j}}\mathbf{PTA}_{d,in} - \frac{\zeta^{j}}{\theta^{j}}\mathbf{Z}_{in} + \nu_{i}^{j} + \mu_{n}^{j} + \varepsilon_{in}^{j}, \qquad (9)$$

where $\boldsymbol{\delta_d}^j = \{\text{shallow, deep}\}, \ \nu_i^j \equiv \ln(\lambda_i^j c_i^j) \text{ and } \mu_n^j \equiv \ln(X_n^j / \sum_{i=1}^N \lambda_i^j \left[c_i^j \kappa_{in}^j\right]^{\frac{-1}{\theta^j}})$ are importer and exporter fixed effects, respectively, and ε_{in}^j is an i.i.d. error term.

The coefficient on tariffs directly identifies the productivity dispersion, $1/\theta^j$. The higher $1/\theta^j$, the stronger the response of trade flows to a cost shifter (here bilateral tariffs). The coefficients on PTAs, $-\frac{\delta_d^j}{\theta^j}$, are expected to be positive. Forming a PTA reduces non-tariff trade barriers, and thus, bilateral trade increases. The coefficient is allowed to vary by sector since non-tariff measures are sector-specific. PTA_{shallow} captures the effect of having a PTA. PTA_{deep} captures the additional effects of having a deep agreement. If $-\frac{\delta_{deep}^j}{\theta^j}$ is not statistically different from zero, in our data, deep trade liberalization does not provide stronger trade cost reductions than a shallow PTA. In our counterfactual analysis, we assume that the TTIP will reduce the costs of non-tariff measures by the

¹⁰In the regressions, the average effect of a PTA is given by the coefficient on the shallow PTA; the effect of the deep PTA is added to this average.

same amount that other PTAs have reduced trade barriers. Thus, we do no not need direct estimates of the levels of NTMs for the 18,000 trade pairs in our analysis,¹¹ nor do we need to speculate about changes in the costs of NTMs that may result from signing the TTIP.

The importer and exporter fixed effects take into account that bilateral trade volumes are influenced by country characteristics. However, the estimates of the PTA dummies could still suffer from an endogeneity bias when, e.g., countries that trade more with each other are also more likely to sign a PTA. In this case, the PTA dummy would overestimate the trade enhancing effect of a PTA. To reduce the endogeneity bias, we use an instrumental variables approach. The instruments should influence the probability to sign a PTA, but other than through the PTA should not affect current trade levels. One such instrument is the contagion index developed by Baldwin and Jaimovich (2012) and, e.g., also used in Martin et al. (2012). It measures the threat of trade diversion country i faces in a trade partner j's market, by counting j's PTAs with third countries weighted with how important the third country's market is for i (i.e. with the third country's share in i's exports). Additionally, we use historical and recent war frequency and lagged average variables for political similarity (average of 2000-2005) as instruments (see Egger et al., 2008, for a brief discussion). The rationale for these variables is that past conflicts will make it harder to negotiate PTAs, while political similarity might make it easier.

Table 1 displays the IV gravity results for the productivity dispersion and the PTA effects for the 17 manufacturing sectors.¹² We drop the 0.5% of outlier observations with the highest tariffs from the sample. In general, our estimations can explain between 60

¹¹We model $134 \times 133 = 17,822$ country pairs.

¹²The estimation is based on trade data from UN COMTRADE. The sample is restricted to the GTAP countries. Data on bilateral tariffs for manufacturing sectors are taken from UNCTAD's TRAINS database. The database can be accessed via the World Bank's World Integrated Trade Solution (WITS) project, https://wits.worldbank.org. The full regression output is relegated to Tables A2 and A3 in the online appendix. We use effectively applied tariffs which are aggregated to the GTAP sector classification with import weights. Other trade cost proxies, i.e., bilateral distance and a dummy for contiguity, are obtained from the CEPII distance database.

and 80% of the variation in bilateral trade volumes. The coefficients on tariffs are highly statistically significant and satisfy the theoretical restriction $\theta^j \in (0, 1)$. The ranking of sectors in terms of their productivity dispersion seems sensible. Sectors like "Mining", "Petroleum", "Ferrous metals" or "Chemicals" which provide fairly homogenous goods have a low θ , i.e. trade flows react relatively strongly to cost changes. Sectors like "Transport equipment" or "Electronics", on the other hand, have a relatively high θ which indicates that they provide relatively heterogeneous sectoral varieties.¹³ We find weak evidence that shallow PTAs increase trade, at least in some sectors. In other sectors, the shallow PTA effect is not statistically different from zero. Deep PTAs, on the other hand, increase bilateral trade quite substantially. For example, in the Processed Food industry, the PTA coefficient is 1.564. This implies bilateral trade increases on average by $e^{1.564} - 1 = 380\%$ in preferential trade agreements. The coefficients on other trade cost proxies are as expected. Distance reduces bilateral trade volumes. A common border, common language and shared colonial past increases trade.¹⁴

Note that some PTA coefficients are not statistically significant. For the simulations, we take the positive PTA coefficients when their t-statistic exceeds the value of one.¹⁵

The estimates obtained for productivity dispersion and their ranking are fairly similar with OLS estimation and also using a simple PTA dummy instead of distinguishing two depths of trade liberalization.¹⁶ The PTA effects obtained from OLS are smaller than the respective IV estimates, in general. This is also true when employing a single PTA dummy and well documented in the literature. Egger et al. (2011), for example, find that unobservable determinants of PTAs typically have a negative effect on bilateral trade

 $^{^{13}}$ The estimates indicate that the agricultural sector has a rather high productivity dispersion. This results from the fact that we have aggregated agriculture, hunting, forestry, and fishing into a single sector.

 $^{^{14}}$ See Felbermayr et al. (2014) for a brief literature survey on the size of existing PTA estimates.

¹⁵In the mining and petroleum sectors, the estimated PTA effects are not stable across different specifications and sometimes implausible high, in spite of evidence from Francois et al. (2009) pointing to the opposite. So we chose to set δ to zero in these two sectors.

¹⁶Full regression output is provided in Tables A6 to A17 in the online appendix.

volumes. Consequently, OLS estimates of PTA dummies are downward biased.

Table 2 provides gravity results for PTA effects in service sectors.¹⁷ In general, our estimations can explain between 90 and 97% of the variation in bilateral service trade flows. Distance seems to be irrelevant for most service sectors. A shared colonial past and a common border, on the other hand, tends to increase service trade. Shallow PTAs do not provide a positive impulse for service trade. The coefficient on the shallow PTA dummy is statistically insignificant in almost all sectors. For the sectors with a statistically significant coefficient, the effect is relatively small. In "Other business service sectors", e.g., a shallow PTA implies about 6% higher trade volumes. A deep PTA, which typically has provisions on service trade, is trade-enhancing in almost all sectors. The effect is in the order of magnitude of 30% in most sectors. Since there are no tariffs levied on service trade flows, we cannot identify θ^{j} in service industries. We instead take an average value from Egger et al. (2012), who estimate a trade cost elasticity for services of 5.959. OLS gravity estimates provide similar results for the service sectors, even though the PTA effects tend to be smaller.¹⁸

3.3 Expenditure and Cost Shares

The remaining parameters, α , β , and γ , and the trade shares π , tariffs τ , and expenditure X are obtained from GTAP. We perfectly match final goods expenditure, the sectoral bilateral trade flows (aggregating intermediate and final goods trade), the cost shares for intermediates and bilateral tariffs. Two adjustments to the data are necessary to align it with the assumptions of the model. The first adjustment relates to different bilateral trade shares between final and intermediate goods trade. The second adjustment concerns the international transport sector.

 $^{^{17}\}mathrm{The}$ full regression output is relegated to Tables A4 and A5 in the online appendix.

 $^{^{18}\}mathrm{The}$ regression output is provided in Tables A8 and A9 the online appendix.

		Ţ	able 1: IV	gravity e	stimates n	nanufactı	uring see	ctors				
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
	Ln ta	riff	$\operatorname{Shallow}$	r PTA	Deep	PTA	Obs.	${ m R}^2$	Weak-ID	Under-ID	Shea's Pa	$rtial R^2$
	-1/ heta	s.e.	$-\delta/ heta$	s.e.	$-\delta/ heta$	s.e.			F-stat	p-value	Shallow	Deep
											PTA	PTA
Agri-Food	-2.861^{***}	(0.636)	1.442^{***}	(0.458)	0.689	(0.630)	7,938	0.607	13.733	0.000	0.022	0.025
Mining	-20.066^{***}	(2.791)	-0.975	(0.793)	4.292^{***}	(1.087)	5,228	0.502	12.853	0.000	0.028	0.027
Food, processed	-3.140^{***}	(0.605)	-0.118	(0.532)	1.564^{**}	(0.712)	8,032	0.663	14.407	0.000	0.021	0.024
Textiles	-4.342^{***}	(0.860)	-0.321	(0.533)	2.357^{***}	(0.700)	8,547	0.744	12.625	0.000	0.020	0.024
Leather	-5.046^{***}	(0.920)	-0.204	(0.650)	2.049^{**}	(0.831)	6,738	0.692	18.096	0.000	0.025	0.029
Wood	-5.684^{***}	(0.806)	1.137^{**}	(0.536)	0.473	(0.707)	7,336	0.711	14.522	0.000	0.026	0.028
Paper	-4.848***	(1.012)	0.826	(0.571)	0.021	(0.762)	7,684	0.732	13.488	0.000	0.022	0.026
Petroleum	-9.538^{***}	(3.335)	-1.702	(1.116)	3.527^{**}	(1.432)	4,101	0.493	6.115	0.000	0.023	0.023
Chemicals	-11.210^{***}	(1.097)	0.194	(0.448)	1.055^{*}	(0.622)	9,027	0.767	14.787	0.000	0.020	0.023
Mineral products	-8.872***	(0.890)	0.430	(0.533)	0.287	(0.692)	7,094	0.732	16.729	0.000	0.023	0.027
Ferrous metals	-10.456^{***}	(1.355)	-0.137	(0.636)	2.245^{***}	(0.845)	5,816	0.659	20.395	0.000	0.028	0.030
Metals nec	-10.756^{***}	(1.901)	1.706^{***}	(0.643)	-0.831	(0.818)	5,498	0.641	11.770	0.000	0.023	0.026
Metal products	-5.725^{***}	(1.170)	0.059	(0.559)	0.859	(0.754)	7,750	0.746	12.459	0.000	0.023	0.026
Motor vehicles	-1.484^{*}	(0.796)	2.210^{***}	(0.608)	-1.175	(0.768)	7,336	0.707	12.255	0.000	0.024	0.030
Electronics	-4.321^{***}	(1.498)	-0.921	(0.617)	2.032^{**}	(0.829)	7,944	0.757	17.779	0.000	0.023	0.027
Machinery nec	-7.186^{***}	(1.022)	0.328	(0.483)	0.550	(0.658)	9,091	0.808	11.217	0.000	0.021	0.025
Manufactures nec	-3.867***	(0.747)	0.526	(0.571)	-0.069	(0.757)	7,811	0.742	20.266	0.000	0.022	0.025
Note: The table show	s gravity estin	lates for nc	n services se	ctors with	importer an	id exporte	r dummie	s and the · ·	e usual gravi	ity controls (not shown).	Standard

statistical significance at the 10, 5 and 1% level, respectively. The dummy for shallow and deep PTAs is classified with data on the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>>= 0, and the deep PTA dummy switches to one for errors (in parametees) are neveroskedasucity-robust. FIA dummes instrumented with single concagion index a la Martin et al. (2012) or parton and Jaimovich (2012), historical and recent war frequency and lagged average variables for political similarity (average of 2000-2005). *, ** and *** indicate depth>= 4. I.e., the shallow dummy gives the effect of having an PTA, the deep dummy gives the additional effect of having a deep agreement.

	(10) al R^2	leep PTA	0.216	0.209	0.215	0.212	0.212	0.212	0.212	0.212	0.212	0.212	0.212	0.212	0.212	0.212
	(9) Shea's Parti	Shallow PTA I	0.198	0.209	0.191	0.186	0.186	0.186	0.186	0.186	0.186	0.186	0.186	0.186	0.186	0.186
	(8) Under-ID	p-value	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
sectors	(7) Weak-ID	F-stat	1002.869	814.976	1026.525	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913	1007.913
service	(6) R^2		0.972	0.964	0.965	0.907	0.959	0.960	0.919	0.957	0.967	0.955	0.964	0.966	0.962	0.943
ble 2: IV gravity estimates	(5) Obs.		11,918	9,161	12,571	12,656	12,656	12,656	12,656	12,656	12,656	12,656	12,656	12,656	12,656	12,656
	(4) (-4)	s.e.	(0.082)	(0.092)	(0.055)	(0.092)	(0.061)	(0.054)	(0.087)	(0.059)	(0.046)	(0.064)	(0.052)	(0.056)	(0.054)	(0.059)
	(3) Deep]	$-\delta/\theta$	0.192^{**}	0.354^{***}	0.250^{***}	0.304^{***}	0.452^{***}	0.267^{***}	0.479^{***}	-0.025	0.076	0.303^{***}	0.136^{***}	0.322^{***}	0.192^{***}	0.268^{***}
T_{i}	(2) PTA	s.e.	(0.048)	(0.056)	(0.030)	(0.061)	(0.038)	(0.031)	(0.056)	(0.039)	(0.027)	(0.039)	(0.031)	(0.036)	(0.033)	(0.035)
	(1) Shallow	$-\delta/ heta$	0.137^{***}	0.025	0.086^{***}	0.024	0.036	0.060*	-0.028	0.064	0.032	0.048	0.038	0.066^{*}	0.030	0.086^{**}
			Electricity	Gas	Water	Construction	Trade services	Transport nec	Water transport	Air transport	Communication	Financial services nec	Insurance	Business services nec	Recreational services	Other services

In the model, the bilateral trade shares are assumed to be identical across use. In the GTAP data, however, bilateral trade shares differ across final and intermediate usage. We match sectoral bilateral trade flows, final goods expenditure shares, and cost shares for intermediates to their empirical counterpart and bilateralize final and intermediate goods trade with the common bilateral trade share. And GTAP has a separate international transportation sector. To match the iceberg trade cost assumption, we assign the international transport margin and its respective share of intermediate demand to the sector demanding the international transportation service. This increases the respective sector's production value. The sectoral value added is calculated as the difference between the so obtained production value and intermediate costs.¹⁹

4 Simulation results: Trade and Welfare Effects of TTIP

We now have paved the way to simulate the effect of PTAs in general equilibrium. The setup allows to explore different scenarios: from a tariffs-only liberalization in selected sectors (e.g., excluding services and agriculture), to a deep PTA encompassing all 32 industries. We apply the framework to the case of the TTIP. We first review some important facts; then we analyze trade creation and diversion. Does TTIP strengthen the transatlantic production network? Then we describe the predicted sectoral value added changes and discuss whether comparative advantage plays a role in shaping sectoral responses to our trade policy shock. Last, we investigate the welfare changes under TTIP. The base scenario is a deep TTIP. However, we also characterize the welfare changes in different scenarios such as deep vs. shallow TTIP, no spillovers vs. spillovers, and we also investigate the interactions of TTIP with a round of multilateral trade liberalization. In

¹⁹This implies that production taxes are part of value added.

the online appendix, we also provide some sensitivity checks pertaining to the parameter estimates, i.e. OLS vs. IV estimation and a single PTA dummy vs. two levels of PTA depth.

4.1 Cross-industry facts for the EU and the US

Tables 3 and 4 provide information on the status quo of trade between the EU and the US for 32 industries. All values are in US dollars and relate to the base year of 2007.²⁰ Column (1) of Table 3 reports for the EU the value added generated in each sector. 73%of total value added (GDP) is generated in the services sectors, 25% in manufacturing, and 2% in agriculture. Columns (2) and (3) shows that total EU exports to the US amount to 465 bn. US dollars; this amounts to about 8% of total exports to non-EU countries. However, in value added terms, the exports of 428 bn. US dollars amount to almost 11% of the total.²¹ This signals that EU exports to the US incorporate relatively little reexports of foreign (including of the US) value added. Column (4) provides tradeweighted sector-level tariff rates that EU exports encounter in the US. Tariffs are low; the average rate (excluding services trade) is just 1.35%. Exports to the world encounter very similar tariff rates; thus, earlier rounds of (multilateral) trade liberalization have not particularly favored EU exports to the US. Columns (5) and (6) report EU exports to the world. This shows that the US is a particularly important market for EU services exporters: in this area the share of exports going to the US usually exceeds the 11%overall average (in VA terms) by a wide margin. The opposite is true in agri-food. The manufacturing industries mostly are below the 11% average, with a few exceptions, most notably Chemicals.

Table 4 provides information for the US side. It shows that the weight of the services

 $^{^{20}}$ This is the most recent year for which input-output data for 134 countries/regions is available. We do not predict baseline values for some future year, as Fontagné et al. (2013) or Francois et al. (2013), since this would introduce additional margins of error.

 $^{^{21}\}mathrm{European}$ value added embodied in US absorption.

	(1)	(\mathbf{n})	(2)	(4)	(5)	(c)	(7)
Castan	(1)	(<i>2</i>)	(3)	(4) • US	(0) E-m ((0)	(1) *Lonlat*
Sector	value	Exp	ports to	0 US Towiffa	Expo	orts to V	voria Taniffa
	added	Gross	VA	1 arms	Gross (in ha	VA UCD)	1 arms
	(111)	on. USI	J)	(111 %)	(in bn.	USD)	(111 %)
Agri-Food	369	2	5	3.43	148	117	1.82
Mining	117	3	9	0.03	65	71	0.00
Food, processed	469	17	9	3.23	262	124	3.12
Textiles	177	7	5	8.08	168	80	8.61
Leather	45	3	2	7.43	48	23	10.57
Wood	94	5	4	0.43	106	51	0.27
Paper	272	5	8	0.01	138	101	0.00
Petroleum	180	16	11	1.26	133	84	0.67
Chemicals	549	71	45	1.13	863	385	1.08
Mineral products	160	6	5	4.76	79	53	2.88
Ferrous metals	109	7	7	0.16	187	81	0.18
Metals nec	56	7	5	1.78	129	44	0.58
Metal products	287	7	13	1.96	154	139	1.38
Motor vehicles	382	65	26	0.86	774	240	0.59
Electronics	146	12	9	0.28	261	100	0.31
Machinery nec	690	80	47	0.81	933	422	0.75
Manufactures nec	129	14	7	0.77	79	40	0.73
Electricity	268	0	8	0.00	28	85	0.00
Gas	21	0	1	0.00	2	6	0.00
Water	40	0	1	0.00	1	7	0.00
Construction	1203	1	5	0.00	43	64	0.00
Trade services	1107	5	12	0.00	129	129	0.00
Transport nec	523	14	20	0.00	106	165	0.00
Water transport	20	0	1	0.00	52	16	0.00
Air transport	80	17	7	0.00	113	46	0.00
Communication	374	5	10	0.00	49	86	0.00
Financial services nec	501	22	22	0.00	114	133	0.00
Insurance	145	21	10	0.00	60	34	0.00
Business services nec	3569	34	90	0.00	438	829	0.00
Recreational services	508	5	7	0.00	66	73	0.00
Other services	2793	14	15	0.00	52	82	0.00
Dwellings	328	0	2	0.00	0	17	0.00
Total	15,709	465	428	1.35	5,780	3,929	1.30

Table 3: Status quo summary statistics EU27

Note: The table shows the EU27's sectoral value added, exports and value added exports to the US and the world in the benchmark year 2007, and the respective tariffs; data source is GTAP 8.1. Tariffs are effectively applied tariffs. Tariffs w.r.t. world are trade weighted-averages. Total tariffs refer to a trade-weighted sum of manufacturing tariffs.

 * EU exports to World exclude intra-EU trade.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Sector	Value	Expo	orts to	EU27	Exp	orts to V	World
	added	Gross	VA	Tariffs	Gross	VA	Tariffs
	(in l	on. USI	D)	(in %)	(in bn.	USD)	(in %)
Agri-Food	187	7	5	5.60	70	37	7.07
Mining	167	4	11	0.00	17	41	0.00
Food, processed	204	4	2	12.01	35	17	10.82
Textiles	77	2	2	7.65	17	9	5.96
Leather	6	0	0	3.95	2	1	6.27
Wood	116	2	2	1.70	11	9	0.74
Paper	212	4	7	0.01	29	29	0.01
Petroleum	27	12	2	1.95	59	6	0.32
Chemicals	351	55	32	2.09	190	104	1.53
Mineral products	66	2	2	3.11	10	9	2.66
Ferrous metals	58	2	4	0.45	19	17	0.21
Metals nec	36	8	5	1.57	35	14	0.87
Metal products	148	5	7	2.13	24	27	1.69
Motor vehicles	247	47	17	3.28	203	63	3.95
Electronics	69	19	5	0.55	95	18	1.40
Machinery nec	474	49	31	1.22	232	128	1.22
Manufactures nec	40	6	3	1.04	20	8	1.10
Electricity	178	0	4	0.00	1	15	0.00
Gas	57	0	1	0.00	1	5	0.00
Water	59	0	1	0.00	0	3	0.00
Construction	882	2	5	0.00	6	19	0.00
Trade services	1804	4	26	0.00	17	96	0.00
Transport nec	259	13	12	0.00	30	37	0.00
Water transport	22	0	1	0.00	2	3	0.00
Air transport	81	11	5	0.00	27	15	0.00
Communication	297	5	6	0.00	11	16	0.00
Financial services nec	1032	22	26	0.00	46	73	0.00
Insurance	296	3	4	0.00	17	16	0.00
Business services nec	1439	38	54	0.00	92	160	0.00
Recreational services	467	7	5	0.00	28	17	0.00
Other services	3187	17	15	0.00	64	55	0.00
Dwellings	1258	0	0	0.00	0	0	0.00
Total	13,800	351	299	2.24	1,412	1,069	2.10

Table 4: Status quo summary statistics USA

Note: The table shows the USA's sectoral value added, exports and value added exports to the EU27 and the world in the benchmark year 2007, and the respective tariffs; data source is GTAP 8.1. Tariffs are effectively applied tariffs. Tariffs w.r.t. world are trade weighted-averages. Total tariffs refer to a trade-weighted sum of manufacturing tariffs.

industries in GDP is even more important for the US than for the EU (82%), but that the agri-food area contributes even less (1%). Moreover, the levels of GDP are similar, with the EU having a slight advantage. The US is more closed; domestic value added embodied in foreign absorption relative to GDP amounts to 8%; in the EU the number stands at about 25%. The EU has a bilateral surplus with the US of 114 bn. USD in gross terms and of 129 bn. USD in value added terms. This signals that EU value added is exported to the US via third countries, e.g., Canada or China. Exports to the EU are relatively more important for the US than exports to the US for the EU. US tariffs appear slightly higher than EU tariffs, but the correlation between the two tariff schedules is relatively high (about 61%).

4.2 Global trade effects of a deep TTIP

Reflecting the official ambitions for the TTIP, our scenario assumes that all transatlantic tariffs are eliminated and costs of NTMs fall to the level observed in other deep PTAs.²² Moreover, we do not allow the agreement to affect trade costs in non-TTIP country pairs, i.e., we abstract from regulatory spillovers. In this case, the volume of international trade increases from 15.3 to 16.4 tn. US dollars²³, i.e. aggregate trade grows by about 7.3%, see also Table 5. Manufacturing trade (about 80% of the initially observed trade volume) increases by 7.8%, trade in agricultural goods even by 12.9%, whereas the growth rate in services trade is only 3.5%. So with TTIP, the share of manufacturing trade increases further (by about 0.46 percentage points) at the expense of the trade in services share which falls by 0.62 percentage points. This seems plausible, since the trade cost reductions are largest in manufacturing.

The impressive trade growth our model predicts might however overstate the value added generated under the TTIP. Trade in intermediates—that are used to produce other

²²We provide sensitivity analysis below.

²³Note that intra-regional trade in GTAP's "Rest of ..." regions is not included in this figure.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
		Expor	ts	Val	ue added	exports	VAX	ratio
	Initial	Share	Growth	Initial	Share	Growth	Initial	With
	(in bn.	(in %)	with TTIP	(in bn.	(in %)	with TTIP	(in %)	TTIP
	USD)		(in %)	USD)		(in %)		(in %)
Agriculture	466	3.0	12.9	478	4.4	5.5	102.5	95.7
Manufacturing	$12,\!154$	79.4	7.9	$6,\!139$	55.9	5.9	50.5	49.6
Services	$2,\!690$	17.6	3.5	$4,\!358$	39.7	6.1	162.0	166.0
Total	$15,\!311$	100.0	7.3	10,975	100.0	5.9	71.7	70.8

Table 5: Global trade effects of a deep TTIP by broad sector

Note: The table shows exports and value added export statistics for agriculture, manufacturing and services on the world level.

traded goods—will lead to a double counting of value added embodied in total trade flows. In the baseline year of 2007, world GDP (or value added) amounts to roughly 53 tn. US dollar. Table 5 shows that about 11.0 tn. US dollars or 20.8% of this value added is generated in one country and ultimately absorbed (i.e. consumed or invested) in another one, i.e. the value added is exported. Comparing recorded trade volumes with value added trade volumes, we find a value added to export (VAX) ratio of 71.7%. So almost 30% of the trade volume measured at the customs is double-counted value added. With the TTIP, our model predicts value added trade to increase by 11.6 tn. US dollar, or about 6%. In value added terms, the world becomes more open with TTIP. About 22% of the value added is exported, an increase of 1.2 percentage points. But, both in the level as well as in the growth created with the TTIP, world value added trade falls behind world trade. The VAX ratio falls by about 1 percentage points to 70.8%. So the TTIP leads to more double-counting of value added in trade statistics, presumably because it fosters the transatlantic production network (with back-and-forth trade of intermediates).

A decomposition of the total value added trade into manufacturing, services and agricultural value added reveals that services trade is more important than recorded trade flows would suggest. Even though manufacturing value added has the highest share in value added transfers between countries, with 56% in total traded value added it lies well below the 80% share of manufacturing trade. The VAX ratio of manufacturing is only 50.5%. This indicates that (1) manufacturing trade partly takes place in the form of intermediates trade and that (2) traded manufacturing goods embody value added of the services industries. Indeed, while the recorded services trade is about 2.7 bn. USD, the trade in embodied services value added is 4.4 bn. USD. So in terms of value added, services account for 40% of value added trade, against 17% in recorded trade volumes. The VAX ratio for services is 162%, implying that service value added trade is by 62% higher than recorded trade. A large fraction of services value added is traded indirectly via the domestic value chain. An example would be domestic accounting or IT services that are embodied in exported cars.

Under the TTIP, agricultural, manufacturing and services value added trade all grow. But whereas recorded manufacturing trade increases by 7.8%, manufacturing value added associated with trade flows only increases by 5.9% (and starting from a lower level). The reverse happens in services sectors, which grow with 6.1% in terms of value added trade, compared to only 3.5% in recorded trade. So, our simulations now suggest that the TTIP will lead to a slight increase in the share of services value added trade of 0.05 percentage points, at the costs of manufacturing (-0.03 percentage points) and agricultural (-0.02 percentage points) value added trade. While the TTIP predominantly fosters manufacturing trade, indirectly, i.e. via domestic supply chains, a lot of the associated value added is generated in the service sectors and not in manufacturing. Given that both liberalizing regions, the US and the EU, are mature economies with a large share of value added in services, this may not come as a surprise. The VAX ratio in agriculture is roughly 103% in the initial situation; with the TTIP it falls to 96%. The US has a comparative advantage in agriculture and increases its agricultural exports. In the US, agriculture uses more services inputs than the average country. This explains why the agricultural VAX ratio goes down with TTIP.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
Region	$ASEAN^b$	Brazil	Canada	China	$EU27^*$	Germany	Mexico	$SACU^a$	Turkey	USA
Export gro	wth (in $\%$) from .	to							
$ASEAN^b$	-1	-0	-2	-3	-1	-3	-3	-0	-0	-7
Brazil	-2	-1	-8	-3	0	0	-7	-2	1	-7
Canada	-1	1	-1	-1	-2	6	-4	1	2	-8
China	-2	1	-2	-1	-1	-5	-3	1	1	-7
$EU27^*$	-6	-5	-9	-6	-1	-5	-7	-4	-5	171
Germany	-7	-6	-9	-6	-8	-1	-7	-3	-4	216
Mexico	0	1	-5	-1	4	-11	-1	2	1	-6
$SACU^a$	-3	-1	-4	-4	2	1	-4	-1	-0	-11
Turkey	-1	-1	-3	-1	-4	-3	-3	-0	-1	-5
USA	-4	-3	-6	-4	212	259	-5	-2	21	-2
Growth of	value adde	ed trans	fers (in %	5) from .	to					
$ASEAN^{b}$	-1	0	-3	-1	1	0	-4	-0	-0	-6
Brazil	-2	-1	-6	-3	0	0	-6	-1	-0	-5
Canada	-0	1	-1	-1	25	42	-4	3	9	-9
China	-1	1	-3	-1	2	1	-3	1	1	-7
$EU27^*$	-6	-5	7	-6	-2	-7	7	-6	-7	120
Germany	-7	-6	12	-7	-10	-2	4	-6	-8	136
Mexico	0	2	-6	0	44	33	-1	5	13	-9
$SACU^a$	-3	-2	-5	-4	-0	-0	-6	-1	-0	-0
Turkey	-2	-2	-1	-3	-6	-4	-2	-2	-1	21
USA	-2	-2	-9	-3	149	169	-8	10	34	-2

Table 6: Aggregate trade effects of deep TTIP

Note: The table shows bilateral changes in trade flows and value added transfers (in %) from deep TTIP. The diagonal describes changes in intra-national trade and/or in the trade volume within a region. * EU27 without Germany. ^a Southern African Customs Union, ^b Association of Southeast Asian Nations.

4.3 Bilateral trade effects of a deep TTIP

While Table 5 offers insights into the evolution of total trade flows under the TTIP, Table 6 looks into its effects on regional trade links. Again, we discuss trade and value added trade changes. Our model predicts a substantial amount of trade creation between the EU and the US in the long run. German and EU exports to the US are expected to roughly triple. And also the US exports to the EU and Germany are expected to go up by 212 and 259%, respectively. However, trade statistics exaggerate the transfer of value added between the two transatlantic economies. German value added exports to the US, e.g., are predicted to increase by 136% instead of 216%. And also the US export growth with Germany is only 169% under the TTIP. One explanation for this observed pattern is the deepening of transatlantic production chains. Then, the US would process more EU value added in the form of intermediate inputs under the TTIP and vice versa. As a consequence, changes in recorded trade flows would overstate how much value added is generated. We will look into this possibility in more detail in the next section.

Figure 2 shows the EU-US trade changes for four sectors: Agri-Food, Chemicals, Transport equipment and Other Business Services. The blue lines represent kernel density estimates for the bilateral trade changes. Trade between the US and EU countries grows in all sectors. The growth is particularly high for Transport Equipment. This is not unexpected, since tariffs are still relatively high in this sector. Figure 2 also shows a kernel density estimate of the respective bilateral growth rate in value added trade between the EU and US for these sectors.²⁴ The two distributions strongly deviate from each other. The pattern is as described for aggregate trade flows. For manufacturing industries, where trade in intermediate inputs is extensive, the growth in value added exports tend to be smaller than the growth in gross exports. For other business services, however, the reverse is true.

We have seen that German exports to the US rise by about 216%. German value added exports, on the other hand, only grow by 136%. This implies that the ratio of value added exports to exports between Germany and the US declines from 0.95 to 0.71. This can mean two things: (1) The domestic value added content of German goods falls with the TTIP. This is confirmed by looking at the inverse Leontief coefficients. The German value added needed to produce output worth one dollar falls between 0.2 and 4.6 US cents²⁵, depending on the sector at hand. (2) Some German value added that is shipped to the US embodied in German exports does not remain in the US, is processed

²⁴We define the value added export of a sector as the value added generated in this particular sector, i.e. in the source sector. It implies that the sector's value added can reach the absorbing country embodied in the trade flows of other sectors as well as directly.

 $^{^{25}}$ To begin with, the German value added content of German (final) goods lies between 51 and 84 US cents per USD of output in the manufacturing sector. In service industries, these values are typically higher and lie between 74 and 96 US cents.



Figure 2: Changes in trade and value added trade in EU-US pairs and within EU

Note: The figure shows kernel density estimates (Epanechnikov, optimal bandwidth) of the changes in trade (blue lines) and value added trade (red lines) for the TTIP scenario of deep trade integration. Trade between EU-US country pairs is displayed in the first column of graphs, within-EU trade is represented in the second column of graphs. The sectors shown are Agri-Food, Chemicals, Transport equipment and Other business services.

further and finally consumed in a third country (e.g. in other EU countries that also intensify their trading relation with the US). In value added terms, this is counted as German value added export to the third country. In all likelihood, the observed decline in the German-US bilateral VAX ratio is a combination of both effects.

Figure 3 shows the sectoral composition and evolution of Germany's exports to the USA under the TTIP. The sectors are arranged in descending order of their trade volumes. The blue, solid line shows the initial export volumes, the red, solid line the counterfactual export volumes with the TTIP. Germany's most important export good to the US is Machinery, followed by Transport Equipment, Chemicals and Other business services. Although the exports of all sectors increase with the TTIP, there is big sectoral heterogeneity. In some sectors, the growth rates are very high—for example, 513% for Transport Equipments or 628% for Electronics—whereas the predicted growth rates are smaller, especially for services industries.

Figure 3 also shows the respective German sectors' value added exports to the US in the initial situation (blue, dashed line) and with TTIP (red, dashed line). Interestingly, in value added terms, the German top export sector is no longer Machinery but Other business services. As discussed earlier, a lot of services value added is traded indirectly. For German business services, for example, the corresponding value added export is roughly 3.5 times higher than the recorded export flow. In these service sectors or also in other upstream sectors, the value added exports growth tends to be substantially higher than the export growth. Again looking at Other business services the export growth is 43%, the respective value added export growth rate, however, is 124%.

In many manufacturing industries, on the other hand, the value added generated by exports to the US is largely exaggerated with export growth statistics. In these sectors, the German value added embodied in exports typically falls quite substantially. For example, in the Transport equipment sector, the initial German value added share is 65 US cents per one USD of output (i.e. 65%). Compared with other industries, this is already



Figure 3: Sectoral composition of German exports and value added exports to the US

Note: The polar plot shows a sectoral decomposition of the German exports (solid lines) and value added exports (dashed lines) to the USA in the initial situation (blue lines) and in the counterfactual with the TTIP (red lines) on a logarithmic scale with the base 10. The sectors are sorted (clockwise) according to the initial value of exports. The difference between the lines gives the growth rate of exports.

low. With the TTIP, this falls by almost 7% and reaches a counterfactual level of 61 US cents per USD of output. Consequently, while these sector's exports grow by 513%, the respective value added export growth is only 367%. In other words, Germany increasingly exports value added generated in the transport equipment sector to the US. However, the export growth rate overstates the amount of value added generated in Germany in this



Figure 4: Sectoral composition of US exports and value added exports to Germany

Note: The polar plot shows a sectoral decomposition of the US exports (solid line) and value added exports (dashed lines) to Germany in the initial situation (blue lines) and in the counterfactual with the TTIP (red lines) on a logarithmic scale with the base 10. The sectors are sorted (clockwise) according to the initial value of exports. The difference between the lines gives the growth rate of exports.

way because (1) the value added is generated in upstream sectors like Other business services and (2) the foreign content of the traded goods increases.

Figure 4 shows a similar graph for US exports to Germany.

Table 6 also shows that the TTIP will lead to substantial trade diversion within the EU.

Intra-EU trade falls by some 5%. In value added terms, the trade diversion is even more

pronounced. Value added exports fall by 7%. Figure 2 provides a more disaggregated view for selected sectors. It shows the kernel density estimates of the trade and value added trade changes within the EU. The distribution of value added changes lies to the left of the distribution of recorded trade changes in all shown sectors. This pattern is stronger for more upstream sectors like Chemicals or Other Business Services. Value added trade also captures indirect effects via sectors that use the sector's output as input. So the pattern of bigger reductions in value added terms can be explained either by the fact that other sectors (in Germany or abroad) reduce the usage of the sector's intermediates or that sectors which heavily rely on the sector's inputs have adverse trade effects or that more of the sector's value added does not stay in the EU trade partner but is processed and shipped on (e.g. to the US), or all of the above.

The TTIP will not only lead to trade diversion within the EU. Both the EU and the US are predicted to export less to non-TTIP countries and mostly also import less from non-TTIP countries. In value added terms, however, the picture is more nuanced. In particular, the predicted patterns show how the growing transatlantic production network interacts with the NorthAmerican production network. Even though the EU exports less to Canada (-9%), Canadian absorption of European value added increases (by +7% for EU27 exclusive Germany, and by even +12% for Germany). A similar pattern emerges for EU trade with Mexico. So, more EU value added is embodied in US goods, and thus absorbed in Canada and Mexico which trade intensively with the US. Vice versa, the EU increases its absorption of Canadian and Mexican value added. It reaches the EU embodied in US exports. Canadian value added exports to the EU27 exclusive Germany, e.g., increases by 25%, and to Germany by 42%; compared to changes of Canadian exports of -2 and +6%, respectively.

The EU, on the other hand, has production networks e.g. with Turkey and South Africa. Even though the SACU exports to the US fall by 11%, the SACU value added exports to the US remain constant (i.e. +0%). EU imports from SACU countries increase,

and so are (partly) indirectly transferred to the US. Another interesting case is the US's trade relationship with Turkey. Turkey is in a currency union with the EU, and so the Turkish import tariffs on US products would fall away with the TTIP. Thus, US exports to Turkey are stimulated and increase by 21%. The asymmetric nature of the Turkish currency union with the EU becomes evident when looking at the Turkish exports to the US. The US does not eliminate its import tariffs on Turkish products with the TTIP, and hence, Turkish exporters suffer from trade diversion on the US market. Turkish exports to the US fall by 5%. But indirectly (presumably via the EU because the input-output linkages between Turkey and the EU are strong), Turkish value added exports to the US grow by 21%.

Last, the TTIP tends to reduce trade flows between third countries. But in some cases, it will lead to trade creation between third countries, such as, e.g., between China and Brazil. This general ambiguity is due to the joint effects of two opposing facts: first, as incomes in third countries fall, demand for imports generally decline; second, trade may rise as exports that went to the EU or the US in the baseline situation are redirected.

Summarizing, trade and value added trade changes mostly follow similar patterns. However, there are some important exceptions where recorded and value added trade flows change nonproportionally or even go in different directions. This is the result of newly forming or existing international production chains.

The described trade changes have important implications for the openness of countries. Our data set allows us to describe changes in countries' openness in value added terms. How does the TTIP affect the amount of foreign value added contained in domestic absorption? In the US, the initial foreign value added share in domestic absorption is 12.8%. With the TTIP, the absorption of foreign value added increases by almost a quarter to 15.8%. So the US becomes more open with the TTIP. In the EU, the foreign value added share in absorption is higher to start with. The EU average is 26%.²⁶ It is lowest in France (21%), and highest in Malta (70%). With a foreign value added share of 26%, Germany lies well within the EU average. With the TTIP, the EU on average increases the foreign value added share by 6.5%. Germany and the UK (both 9%) experience the largest increases of foreign value added absorption. On the other end of the within-EU distribution are Slovenia, Estonia and the Czech Republic with an increase of below 0.5%.

While the foreign value added in absorption increases in the TTIP countries, it falls by 1.3%, on average, in non-TTIP countries. The reductions are largest for Canada (-3.1%), Mexico (-2.6%), and Japan (-2.4%); while in some African, non-EU European and Central Asian countries the foreign value added in domestic absorption increases slightly. So, in general, non-TTIP countries tend to become less open.

4.4 Bilateral trade balances

Missing an intertemporal savings-investment tradeoff, the model fixes multilateral trade imbalances to the initial situation. However, in the counterfactual situation, the structure of trade imbalances will change. With the TTIP, the US increases its trade deficit with the EU from 118.4 bn. USD (of which 43.5 with Germany) to 179.0 bn. USD.²⁷ This is a long-term increase of more than 40 bn. USD. The US deficit with the SACU region and the oil exporting countries also increases. On the other hand, the US deficit with most other regions is reduced. For example, the deficit with Canada is reduced by 9.4 bn. USD, the one with the Pacific Alliance countries by 6.3 bn. USD. Furthermore, the deficit with Asian countries, in particular China (+22 bn. USD), East Asia (+18 bn. USD), ASEAN (+7 bn. USD) and South Asia (+4 bn. USD) is cut substantially. The same pattern emerges for the US bilateral deficit in value added transfers.

²⁶This is a weighted average, where the country's total absorption is used as weight.

²⁷To construct the trade balances, imports are valued inclusive of tariffs.

Germany increases its trade surplus to the US by 46 bn. USD after the introduction of TTIP (from 34.4 to 80.3 bn. USD). The trade deficit with East Asia and also with China is reduced, albeit by 124 and 295 mio. USD only. With most other regions, Germany has a trade surplus. This surplus is cut substantially with the other EU27 countries (-29 bn. USD) and also with most other regions. In value added terms, the German surplus with the US increases less (42 instead of 46 bn. USD). So the regular trade surplus overstates the increasing imbalance.

4.5 The transatlantic production network

In this section, we analyze the transatlantic production network in more detail. We focus on the trade relation between the two biggest Atlantic economies, Germany and the US. To capture the effects along the total world production chain, we focus on value added flows. The Leontief logic allows to single out for each final goods trade between two countries, how much value added is generated in each source country and sector. Thus, the production networks can be analyzed coming from two different angles (see also Aichele and Heiland, 2014). First, we can ask through which final goods producing countries German value added reaches the US. Second, we can ask from which regions the value added stems that Germany processes into final goods exports to the US. The first method investigates the (final) route German value added takes into the US, while the second method highlights the sourcing structure of the German economy.

German value added exports to the US rise by 136%. Table 7 summarizes via which final goods producing regions, i.e. processing region, the German value added reaches the US consumers. Column (1) shows that, in the initial situation, almost 40 bn. USD of German value added reaches the US embodied in German final goods exports to the US. An even more substantial part reaches the US via products that are finally assembled in the US (roughly 50 bn. US dollars). This captures that Germany may export intermediates to the US or other countries, which in turn export final or intermediate

	(1)	(2)	(3)	(4)	(5)	(6)
Processing region	V	A flow (in Mio US	SD)	VA flo	w, growth ra	ate (in %)
	Initial	With TTIP,	With	Scale	Sourcing	Total
		initial sourcing	TTIP	effect	effect	effect
		structure				
USA	52,588	49,894	$106,\!270$	-5	113	102
Germany	$38,\!175$	121,985	$115,\!867$	220	-5	204
EU27	6,506	$23,\!285$	$19,\!807$	258	-15	204
China	1,826	$1,\!679$	$1,\!585$	-8	-6	-13
East Asia	1,503	$1,\!192$	$1,\!149$	-21	-4	-24
Canada	$1,\!311$	965	1,216	-26	26	-7
Pacific Alliance	1,065	886	967	-17	9	-9
$ASEAN^d$	976	901	868	-8	-4	-11
EFTA^{b}	747	703	672	-6	-4	-10
$MENA^{c}$	401	377	351	-6	-7	-12
South Asia	167	149	139	-11	-7	-17
MERCOSUR	114	94	90	-17	-5	-21
Eurasian Customs Union	81	77	72	-5	-6	-11
$SACU^{a}$	70	57	54	-18	-6	-23
Latin America & Caribbean	66	60	64	-10	7	-3
Australia & New Zealand	57	54	50	-6	-6	-11
Turkey	44	37	34	-15	-8	-21
Oil exporters	40	40	38	0	-5	-5
Sub-Saharan Africa	36	35	33	-3	-6	-9
Central Asia	28	25	24	-9	-6	-15
Rest of Europe	27	25	24	-8	-5	-13
Rest of World	16	15	14	-7	-6	-13
Oceania	5	5	5	-0	-8	-8

Table 7: German value added flows to US via different processing regions

Note: The table shows German value added flows to the US via different processing regions, i.e. final goods exporters. Scale effect: VA growth going from initial to deep TTIP, but keeping sourcing structure (i.e. Leontief coefficients) as in initial situation. The scale effect corresponds to the export growth of the processing region with the US. Sourcing effect: VA growth going from deep TTIP with initial sourcing structure to deep TTIP with new sourcing structure (i.e. counterfactual Leontief coefficients). It shows the effect of changes in the supply chains. Total effect: total VA growth going from initial to deep TTIP.

 a Southern African Customs Union, b European Free Trade Association, c Middle East & North Africa, d Association of Southeast Asian Nations.

goods for further processing to the US. Under the same logic, other EU countries' exports also contain German value added of around 6.5 bn. USD. TTIP now leads to intensified transatlantic trade. German exports to the US roughly triple; those of other EU27 countries, on average, increase by 170%. Keeping the sourcing structure constant (i.e. fixing the Leontief inverse coefficient in the baseline and only increasing final goods trade flows to their counterfactual value), this scale effect implies that more German value added
reaches the US via German exports, but also via final goods exports of other EU countries to the US. The scale effect also shows trade diversion; the US trades less with all non-EU regions including itself, and this implies less German value added (that is used to produce the other regions' goods) reaches the US via other regions. The intra US trade also falls by about 6%. But in GE, the sourcing structure adjusts with the TTIP. The US has more imports from the EU, and in part uses them as inputs to produce its own goods. Therefore, the US sourcing effect in column (5) is positive. More German value added reaches the US consumers embodied in US final goods because they use more German, and generally EU inputs (which also contain considerable amounts of German value added). The sourcing effect is negative for most other regions. Trade diversion implies that non-TTIP regions use less TTIP inputs to produce their output. However, Canada, the Pacific Alliance and Latin America display a positive sourcing effect. These regions all trade a lot with the US and are an integral part of the North American production network. Since the US uses more German (and EU) products as inputs, these countries indirectly also source more value added from Germany via the US.

Overall, German value added increasingly reaches the US via final goods exports of Germany and the other EU countries as well as intra-USA trade. In other words, the TTIP strengthens the transatlantic production network.

Table 8 looks into the sourcing structure of the German economy. Initially, roughly 52 bn. USD of value added reach the US consumers via final goods exports of Germany. The largest share, about three quarters, is value added generated in Germany itself. This indicates that the domestic value chain is still predominant. Other EU countries, the USA, East Asia, EFTA countries and China are also important source countries. The substitution and income effect leads to more US final goods imports from Germany. Thus, keeping the sourcing structure constant, the value added sourced in all regions and processed in Germany for US consumers increases. The effect is not uniform across all source countries due to sectoral heterogeneity. This depends on how important the source

	(1)	(2)	(3)	(4)	(5)	(6)
VA source country	V	A flow (in Mio US	SD)	VA flo	w, growth r	ate (in %)
	Initial	With TTIP,	With	Scale	Sourcing	Total
		initial sourcing	TTIP	effect	effect	effect
		structure				
Germany	$38,\!175$	121,985	$115,\!867$	220	-5	204
EU27	7,022	$27,\!887$	23,791	297	-15	239
USA	$1,\!163$	$4,\!655$	$15,\!691$	300	237	1249
East Asia	867	$3,\!483$	$3,\!142$	302	-10	263
EFTA^{b}	778	$2,\!634$	$2,\!401$	239	-9	209
China	744	2,741	2,725	269	-1	266
Eurasian Customs Union	717	2,487	2,321	247	-7	224
ASEAN^d	340	1,273	$1,\!224$	275	-4	260
$MENA^{c}$	329	1,169	$1,\!145$	256	-2	248
MERCOSUR	230	898	880	291	-2	283
Pacific Alliance	229	965	1,097	321	14	379
Oil exporters	224	834	846	273	1	278
Turkey	175	780	650	345	-17	271
South Asia	136	544	538	301	-1	296
Canada	128	500	781	290	56	509
Sub-Saharan Africa	112	410	441	266	8	294
$SACU^{a}$	110	424	389	284	-8	252
Central Asia	108	387	361	257	-7	232
Australia & New Zealand	94	346	335	269	-3	258
Rest of Europe	62	240	212	287	-12	241
Latin America & Caribbean	54	207	219	286	6	310
Rest of World	17	66	65	277	-1	272
Oceania	8	31	29	261	-5	242

Table 8: Value added flows to US with final processing in Germany

Note: The table shows by source country the value added flows embodied in German final goods exports to the US. Scale effect: VA growth going from initial to deep TTIP, but keeping the (German) sourcing structure (i.e. Leontief coefficients) as in initial situation. The scale effect corresponds to the export growth of Germany, but due to sectoral heterogeneity the scale effect can vary by source country. Sourcing effect: VA growth going from deep TTIP with initial sourcing structure to deep TTIP with new sourcing structure (i.e. counterfactual Leontief coefficients). It shows the effect of changes in the supply chains. Total effect: total VA growth going from initial to deep TTIP. ^a Southern African Customs Union, ^b European Free Trade Association, ^c Middle East & North Africa, ^d Association of Southeast Asian Nations.

countries are in faster growing sectors. When allowing for the adjustment of the inputoutput coefficients along the global value chain, the US considerably increases its role for Germany as a source of value added. Its share in processed value added goes up from 2 to 9%. Countries in the North American production network (like Canada or Mexico) also benefit from this. Germany sources more value added from these countries (because they trade intensively with the US). However, their shares still remain small. Summarizing, the transatlantic production network is strengthened with the TTIP. Germany relies more on US inputs to process its output. And German value added becomes more important in the absorption of US consumers directly (by final goods exports), and indirectly by exporting intermediates, in particular to the US and other EU countries, which then process the German value added for final consumption in the US.

4.6 Effects on sectoral value added

In this section, we investigate the effects of the TTIP on the production structure. All results stem from the base scenario, i.e. deep TTIP from IV regressions. Table 9 shows by region the share of agriculture, manufacturing and services in total value added. In both the EU and the US, services industries make up by far the largest part of total value added. 73% in the EU and 82% in the US. Manufacturing is still more important in the EU than in the US. In Germany, the share of manufacturing is 28.4% and higher than in most developed economies. Agricultural value added, on the other hand, plays a minor role. Our simulations predict that the TTIP leads to a slight reindustrialization in the EU. The share of manufacturing value added in total value added increases by 0.3 percentage points, on average. In Germany, it is even 0.4 percentage points, and starting from a higher initial value. On the other hand, both service industries and agriculture lose a small share in total value added.

Within the EU, however, this trend of reindustrialization is not universal. Figure 5 shows a kernel density estimate of the changes in the manufacturing (dark blue lines) and service industry (light blue lines) share. The distribution for the TTIP countries (solid line) of manufacturing peaks slightly above zero, but also reaches into the negative area. So while the shifts are not dramatic, some Central and Eastern European economies—like Bulgaria, the Czech Republic, Estonia, Poland and Slovenia—and also Spain and Lux-embourg shift away from manufacturing. In Bulgaria, the agricultural share increases. In the other countries, service industries pick up. Still, in the big economic centers in

1able 9.	THE ENEC	<u>, 15 01 a 1 1 11</u>	UII SECI	orar value au	lueu	
	(1)	(2)	(3)	(4)	(5)	(6)
	Ag	griculture	Mar	ufacturing	S	Services
	Initial	Counter-	Initial	Counter-	Initial	Counter-
	share	factual	share	factual	share	factual
	(in %)	change	(in %)	change	(in %)	change
		(in % points)		(in % points)		(in % points)
TTIP countries						
EU27	2.3	-0.1	24.6	0.3	73.1	-0.2
Germany	1.4	-0.1	28.4	0.4	70.2	-0.3
USA	1.4	0.1	16.6	-0.1	82.0	-0.0
Non-TTIP regions						
ASEAN	10.6	-0.0	36.7	-0.2	52.7	0.3
Australia & New Zealand	4.0	-0.0	19.8	-0.1	76.1	0.1
Canada	2.2	-0.0	23.0	-0.2	74.8	0.2
Central Asia	9.3	-0.0	28.9	-0.1	61.8	0.1
China	10.8	-0.1	37.6	-0.0	51.6	0.1
EFTA	1.4	-0.0	24.4	-0.2	74.2	0.2
East Asia	1.5	0.0	21.8	-0.2	76.7	0.1
Eurasian Customs Union	6.9	-0.0	27.8	-0.1	65.3	0.1
Latin America & Caribbean	8.3	-0.1	20.6	-0.1	71.1	0.2
MERCOSUR	7.2	-0.1	23.0	-0.0	69.8	0.1
MENA	8.0	-0.0	39.3	-0.1	52.6	0.2
Oceania	5.8	-0.1	25.7	-0.3	68.5	0.4
Oil exporters	0.9	-0.0	62.1	-0.2	37.0	0.2
Pacific Alliance	5.1	-0.0	29.1	0.0	65.8	-0.0
Rest of Europe	7.7	-0.0	24.6	-0.2	67.6	0.2
Rest of World	11.7	-0.1	14.2	-0.3	74.1	0.4
South Asia	17.3	-0.0	19.5	-0.1	63.1	0.2
SACU	3.5	-0.0	25.8	-0.1	70.8	0.1
Sub-Saharan Africa	26.2	-0.1	33.4	-0.0	40.4	0.1
Turkey	7.6	-0.1	27.2	-0.1	65.2	0.2

Table 9: The effects of a TTIP on sectoral value added

Note: The table shows the initial shares of agriculture, manufacturing and service industries in total value added in % (odd columns), and the respective counterfactual change with deep TTIP in percentage points (even columns).

Europe—like Germany, Italy, the UK and also France—there is a shift towards manufacturing under the TTIP. In the US, on the other hand, the manufacturing share declines by 0.08 percentage points and also services industries lose some ground to the benefit of agriculture.

Table 9 also shows that in non-TTIP countries, the opposite trend occurs. The share of services in total value added tends to increase at the detriment of manufacturing. This pattern is also support by Figure 5. The distribution of the changes in manufacturing shares for non-TTIP countries (dashed, dark blue line) is mostly in the negative area.



Figure 5: Changes in manufacturing and services shares with the TTIP

Note: The figure shows the density of changes of the share of manufacturing (dark blue line) and services (light blue line) value added (in percentage points) for the TTIP scenario of deep trade integration distinguished by the TTIP (solid line) and non-TTIP countries (dashed line). Malta for scaling reasons not in graph.

Next, we provide a detailed analysis of the sectoral changes in the biggest Atlantic economies, namely Germany and the USA. Figure 6 shows the changes in sectoral value added from the TTIP (red lines, right scale). As a reference, the figure shows the initial value added of all sectors with a value added share above 1% in descending order of their share in total value added (dark blue bars, left scale).

The effect of the TTIP on sectoral production values is very heterogenous. In Germany, the transport equipment sector experiences the largest growth in value added, +17%. Given that tariffs on transport equipment are still relatively high in the US and that German car manufacturing is an important economic sector (about 4% of total German value added), this seems plausible. Sectors that provide services to the car manufacturing industry also grow. For example, Other Business Services have a value added growth of about 0.6%, although from a larger base (the sector makes up about a quarter of total



Figure 6: Sectoral value added: initial levels and TTIP-induced changes

Note: The figure shows the changes in sectoral output (in %) for the TTIP scenario of deep trade integration, for (a) Germany and (b) the USA. The figure only lists sectors with a value added share of above 1%. Due to the Cobb-Douglas production structure, output changes are equivalent to value added changes. The sectors are listed on the vertical axis with their GTAP short description. An overview of the sectors is provided in Table A1 in the online appendix.

value added in Germany). The chemical and machinery sector, on the other hand, shrink.

In the US, the most important sectors are service industries like Other, Trade, Business, and Financial Services and Construction. These sectors slightly grow with the TTIP. As in Germany, however, the transport equipment sector grows most, by about 10%. The US can also increase its production of Agri-Food products. Interestingly the Processed Food sector shrinks.

The observed pattern suggests that comparative advantage plays a role for specialization under the TTIP. Germany specializes more in manufacturing, and especially so in the transport equipment sector. The US, on the other hand, presumably has a comparative advantage in chemicals and agriculture and increases its share of agriculture in total value added. A regression of the change in the (multilateral) revealed comparative advantage²⁸ (RCA, in percentage points) on the initial RCA gives a positive coefficient of 0.047, statistically significant at the 1% level.²⁹ So the increase in the RCA is positively correlated with the initial revealed comparative advantage under the TTIP. This suggests that, with the TTIP, countries tend to specialize in sectors they have a comparative advantage in.

Figure 7 shows that in Germany, the transport equipment sector is the only sector with an initial revealed comparative advantage that can further increase its RCA. Other sectors, where Germany is relatively good at exporting, like "Chemicals" or "Machinery", lose some of their revealed comparative advantage. The upper left quadrant shows sectors, for which Germany so far had a revealed comparative disadvantage, and could increase its competitiveness. Examples are "Electronics", "Textiles" or "Leather". However, these sectors might indirectly benefit from increased transport equipment exports. So while there is some evidence, that Germany specializes in comparative advantage industries,

²⁸The RCA of a country in a given sector is defined as the sector's share in the country's export in relation to the sector's share in world exports (see Balassa, 1965). An RCA larger than 1 reveals that, compared to the world, the country exports relatively much of the sector's goods, i.e. the country has a revealed comparative advantage. And vice versa for an RCA smaller than one.

 $^{^{29}}$ The result does not change when including sector dummies or looking at the subsample of sectors with an RCA larger than one.



Figure 7: Changes in revealed comparative advantage

Note: The figure shows a scatter plot of the sectoral measure of revealed comparative advantage plotted against its change (in percentage points) from switching to the TTIP scenario with deep integration, for (a) Germany and (b) the United States.

	(1)	(2)	(3)	(4)	(5)	(6)
			$\ln(\widetilde{V}$	(\widehat{A})		
Ln price change	0.143*	0.140*	0.135^{*}	0.025	0.165^{**}	0.057
	(0.078)	(0.078)	(0.078)	(0.091)	(0.082)	(0.093)
Ln productivity dispersion	-0.123***	-0.121***	-0.122^{***}		-0.112***	
	(0.024)	(0.024)	(0.024)		(0.024)	
Ln value added cost share	0.044^{***}	0.060^{***}	0.057^{***}	-0.024	0.059^{***}	-0.018
	(0.015)	(0.016)	(0.016)	(0.022)	(0.015)	(0.023)
Ln initial VA share	0.013	-0.039**	-0.060***	0.012	-0.068***	-0.020
	(0.016)	(0.016)	(0.017)	(0.036)	(0.017)	(0.038)
Ln initial RCA		0.078^{***}	0.089^{***}	0.052^{***}	0.079^{***}	0.048^{***}
		(0.015)	(0.016)	(0.016)	(0.017)	(0.015)
Ln initial import penetration			-0.048***	0.060***	-0.062***	0.026
			(0.014)	(0.019)	(0.013)	(0.019)
Sector FE				YES		YES
Country FE					YES	YES
Observations	4,266	4,132	4,132	4,132	4,132	4,132
\mathbb{R}^2	0.050	0.053	0.055	0.140	0.110	0.187

Table 10: Explaining sectoral value added change

Note: The table shows standardized coefficients of estimates on pooled sectoral data of the log of \widehat{VA}_i^s . Standard errors (in parentheses) are heteroskedasticity-robust. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The covariates are (all in logs): the sectoral productivity dispersion θ^s , the sectoral value added intensity β_i^s , the initial share of the sectoral value added in a country's total value added (to measure the sector's importance) $\frac{VA_i^s}{VA_i}$, the initial revealed comparative advantage (RCA), the initial level of import penetration $1 - \frac{Imp_{ii}}{\sum_j Imp_{ji}}$, and the initial export share $1 - \frac{Exp_{ii}}{\sum_j Exp_{ij}}$.

there are many rules to the exception. I.e. there are many observations in the off-diagonal quadrants in Figure 7.

For the US, a similar picture emerges. The country increases its revealed comparative advantage in three of its six RCA industries, namely "Transport equipment", "Agriculture" and "Chemicals". It also gains ground in some revealed comparative disadvantage industries like "Electronics".

Next, we look into the determinants of sectoral value added changes. Table 10 shows standardized coefficients of pooled sectoral regressions of $\ln(\widehat{VA}_i^s)$ on different explanatory variables.³⁰ Some regressions include sector and/or country dummies. Between 5 and 20%

 $^{^{30}}$ From column (1) to (2), the number of observations drops by 32 because there is no trade in the sector 32 "Dwellings" and therefore, the RCA measure is not defined.

of the variation in sectoral changes of value added can be explained with our regressions. Aside from country or sector dummies, a sector's productivity dispersion has the highest explanatory power for sectoral changes in value added. A one standard deviation increase of $\ln \theta$ leads to a reduction of $\ln VA$ of about 0.12 of a standard deviation. In highly dispersed sectors, i.e. sectors with a high θ , output reacts less to our trade policy shock.³¹ In more dispersed sectors sectoral varieties are less substitutable. Consequently, trade cost changes have a smaller impact on sectoral trade and sectoral output. The second most important explanatory variable is revealed comparative advantage. Sectors with a higher initial revealed comparative advantage experience stronger growth in sectoral value added. So with the TTIP, countries tend to specialize in sectors in which they have a comparative advantage. Higher import penetration, on the other hand, deters sectoral growth. Labor-intensive sectors have higher value added growth. A sector's initial share in total value added is negatively related to sectoral growth. However, the coefficient is not statistically different from zero in most cases. So there is some weak evidence, that countries do not necessarily grow in important sectors. An alternative explanation for this finding is that growth rates are smaller in large sectors because they start from a higher initial output level.

4.7 Welfare effects of the TTIP

In our baseline scenario, we assume that the TTIP will be a deep PTA. Given that most new PTAs between developed countries are PTAs of the new generation and go way beyond a mere reduction or elimination of bilateral tariffs, this choice seems reasonable. Moreover, the negotiating mandate that EU member states have given to the EU Commission makes the high level of ambition for the agreement very clear. Nonetheless, we contrast the so obtained results with alternative TTIP scenarios in the next section. In a

³¹Note that, since productivity is sector-specific but does not vary across countries, we cannot identify the coefficient on sectoral productivity dispersion once we include sector fixed effects.

first set of scenarios, the depth of trade liberalization is adjusted. In a second set of scenarios, we investigate the effects of regulatory spillovers. Last, we investigate the welfare effects when the TTIP is accompanied by a round of multilateral tariff reductions under the WTO.

A regional breakdown of the effect of a deep TTIP on countries' real incomes is provided in column 1 of Table 11.³² With the TTIP, the US real income increases by 2.7%. In the EU, real income rises slightly less on average, by 2.1%.³³ The effect is dispersed within the EU. While the core EU countries like the UK, Germany and France benefit more than the average EU country, in particular the Southern European countries like Spain, Italy and Greece and most Central and Eastern European countries (CEEC) only realize welfare gains of around 1%. I.e. the TTIP leads to a divergence in real incomes within the EU. Hence, the finding in Felbermayr et al. (2014) does not seem to be robust to allowing for multiple sectors and input-output links.

Most non-TTIP regions also slightly benefit from the TTIP. Regions close to the EU, like the Rest of Europe, Central Asia, the EFTA and MENA countries as well as Turkey stand to gain most. These regions are deeply embedded in the European production network and therefore benefit from an increased demand for (final and intermediate) goods in the EU. For Canada, the increase in real income is small (about 0.1%). Countries in the Asian production network, on the other hand, tend to lose from the TTIP. Our simulations predict a reduction of real income of 0.23% for China, 0.20% for ASEAN countries and 0.13% for Japan and South Korea (East Asia). Since the TTIP leads to losses in the big economies in East and Southeast Asia it is, on average, marginally detrimental for non-TTIP countries. Given that these losses will be realized in the medium run (over 10-12 years), the annualized malus on growth is relatively low compared to the annual growth rates in most developing countries. So, taken together, TTIP is beneficial for the

³²The online appendix presents effects for all 134 countries/regions.

 $^{^{33}}$ The EU countries' real income changes are weighted with GDP shares to calculate the average.

	(1)	(2)	(3)	(4)	(5)	(6)
			Welfare gr	rowth (in	n %)	
Scenario:	Base	Tariffs	Shallow	Spil	lovers	TTIP
		only	TTIP	direct	indirect	+ Doha
TTIP	2.37	0.00	1.79	2.88	2.98	2.49
USA	2.68	0.01	2.06	3.25	3.37	2.77
EU27	2.12	0.00	1.57	2.57	2.65	2.26
Non-TTIP	-0.03	-0.00	0.05	0.23	1.21	0.44
Rest of World	0.40	0.00	0.36	0.65	3.37	0.57
Rest of Europe	0.29	0.01	0.28	0.74	2.16	0.46
Central Asia	0.20	-0.00	0.19	0.50	2.38	0.21
Eurasian Customs Union	0.17	-0.00	0.13	0.28	1.59	0.89
$SACU^{a}$	0.15	-0.00	0.14	0.38	1.85	0.67
EFTA^{b}	0.15	-0.00	0.19	0.55	1.70	0.48
Canada	0.12	-0.00	0.13	0.49	1.60	0.19
$MENA^{c}$	0.12	-0.00	0.14	0.38	1.87	0.65
Latin America & Caribbean	0.11	-0.00	0.11	0.38	1.54	0.44
Sub-Saharan Africa	0.10	0.00	0.12	0.37	1.95	0.81
Turkey	0.10	-0.01	0.10	0.35	1.16	0.30
Oil exporters	0.09	-0.01	0.19	0.45	2.35	0.71
Australia & New Zealand	0.08	-0.00	0.08	0.20	1.12	0.55
Oceania	0.06	-0.00	0.09	0.34	1.94	-0.02
South Asia	0.06	-0.00	0.09	0.23	0.98	1.25
MERCOSUR	0.00	-0.00	0.03	0.18	0.84	0.20
Pacific Alliance	-0.07	-0.01	-0.02	0.25	1.12	-0.16
East Asia	-0.13	-0.00	-0.08	0.01	0.48	0.32
$ASEAN^d$	-0.19	-0.01	0.05	0.28	2.15	0.42
China	-0.23	-0.01	0.05	0.20	1.14	0.41
World	1.32	0.00	1.02	1.72	2.20	1.60

Table 11: Simulated changes of real income due to the TTIP

Note: The regional real income change is a GDP-weighted sum of the respective countries' real income changes. An overview of the classification of countries into regions is provided in the online appendix. Base: deep TTIP scenario. Direct spillovers: deep TTIP scenario plus trade cost reduction for non-TTIP countries' exports to the EU and US by 20% of the trade cost reduction within TTIP. Indirect spillovers: Direct spillover scenario plus trade cost reduction between third countries and TTIP countries' exports to third countries of 10% of the trade cost reductions within TTIP. TTIP+Doha: deep TTIP scenario plus elimination of all tariffs between WTO member states.

 a Southern African Customs Union, b European Free Trade Association, c Middle East & North Africa, d Association of Southeast Asian Nations.

world. It leads to an increase in world real income of 1.3%, on average.

4.8 Welfare effects of TTIP—Alternative scenarios

To get a better sense of the magnitude of the effects, we now vary the intensity of trade cost reductions the TTIP may provide. First, the TTIP may only lead to the elimination of tariffs. Second, it may only achieve a shallow instead of a deep reduction of NTMs. The results are presented in columns 2 and 3 of Table 11. The tariffs between the EU and the US are already relatively low. An elimination of the bilateral tariffs therefore does not lead to any measurable gains from trade. If the negotiated TTIP agreement is shallow the gains from trade fall slightly short of the ones in the baseline scenario. However, the expected welfare gains are still sizable, 2.1% higher real income for the US and 1.6% for the EU, on average. The negative consequences for third countries are also less severe. Non-TTIP countries slightly raise their real income by 0.05%, on average.

Next, we discuss two scenarios where the TTIP also leads to spillovers. The rationale behind these scenarios is that the TTIP might lead to unified transatlantic product standards due to regulatory cooperation, or the mutual recognition of standards (see, e.g., the discussions in WTO, 2012). This potentially leads to trade cost reductions for firms who export to the US and the EU (direct spillovers) because they only have to abide by one standard for exports to both regions.³⁴ If the TTIP shapes global product standards, the trade cost reductions may even spill over to trade between third countries (indirect spillovers).³⁵ Whether spillovers are likely is not clear. So far, there is little empirical evidence pointing toward positive third-country effects of mutual recognition agreements or standard harmonization (see Chen and Mattoo, 2008; Baller, 2007). And if so, it seems to be crucial that the countries can adopt the (high) standards. So especially developing countries are hurt by standard harmonization.³⁶ Therefore, we use rather ad-hoc assump-

 $^{^{34}\}mathrm{For}$ the simulation, we work with a trade cost reduction of 20% of the full PTA effect that arises between the TTIP countries.

 $^{^{35}\}mathrm{We}$ assume that the indirect spillovers amount to 10% of the trade cost reduction between the TTIP countries.

³⁶Cadot et al. (2012) show that regulatory cooperation in North-South PTAs deters South-South trade.



Figure 8: Density of changes in real income with the TTIP

Note: The figure shows the density of changes in real income (in %) for different TTIP scenarios.

tions as to how spillovers affect trade costs. Nevertheless, it is instructive to simulate spillover scenarios. In terms of policy implications, they provide interesting insights, as we will discuss later.

Direct spillovers from unified standards bring additional welfare gains for the TTIP regions (see column 3 of Table 11). For the EU and the US, the real income increase is by 0.4 and 0.6 percentage points, respectively, higher than in the baseline. But also non-TTIP countries benefit from 0.3 percentage points higher growth. If the gains from global standards also indirectly spill over to third country trade, non-TTIP countries will also largely benefit from the TTIP, see column 4 of Table 11. The average real income increase is 1.2%. The additional effects for the TTIP countries are small and in the magnitude of 0.1 percentage points more growth.

Figure 8 visualizes the welfare effects of the TTIP across different scenarios. It shows

the density of the distribution of real income changes. The dark blue line represents the baseline scenario. It has two peaks, one around zero which captures the real income effects for non-TTIP countries and a second peak to the right (around 2%) which traces the effects on TTIP countries. With a shallow TTIP (light blue line), the two peaks slightly move together. The effects for both TTIP and non-TTIP countries are dampened. With direct spillovers (orange line), the distribution moves to the right. All countries tend to be better off and there are virtually no negative real income effects. What is more, the shift of the first peak is relatively larger. So non-TTIP countries benefit more from spillovers. Nevertheless, the TTIP countries would also benefit from unified product standards.

These findings imply that a TTIP with spillovers is a win-win situation for the TTIP insiders as well as outsiders. Consequently, an inclusive TTIP that, e.g., interprets rules of origins fairly generously can increase good-will in third countries. This in turn may stimulate multilateral trade liberalization. But of course it all hinges on the (implicit) assumption, that developing countries can meet the set standards.

Last, we simulate the effects of a TTIP in a world where a further wave of multilateral tariff liberalizations could be achieved in the Doha round. More specifically, the simulation is based on eliminating all tariffs between WTO member states. If the TTIP is inclusive, i.e. the agreement is such that trade diversion in third countries is limited as best as possible, it may help to stimulate another round of multilateral tariff liberalization. Indeed, the international economics literature discusses whether PTAs constitute a "building block" or "stumbling block" for free trade (see Bhagwati, 1991, for an introduction of the concepts). In the TTIP + Doha scenario, we find that real income rises for all regions except Oceania and the Pacific Alliance. The additional effects for the TTIP countries are relatively small. However, for some regions like South Asia, Sub-Saharan Africa or the Eurasian Customs Union a sizeable positive impulse for the economy can be expected. These regions are still characterized by considerable MFN tariffs.

Summarizing, only a deep TTIP will reap the full gains from trade for TTIP countries.

They are in the order of magnitude of 2 to 3% real income growth. If the TTIP is accompanied by spillover effects, adverse effects for other regions of the world can be prevented and instead turned into small welfare increases as well. This is also beneficial for TTIP countries. Especially from the point of view of developing countries, it is highly desirable to accompany the TTIP by another round of multilateral trade liberalization. This will raise real income in all regions as well.

5 Conclusion

A number of systemically relevant preferential trade agreements (PTAs) are under negotiation. The largest of these mega-regionals is the proposed Transatlantic Trade and Investment Partnership (TTIP). In this paper, we have extended the recent multi-country, multi-industry Ricardian trade model by Caliendo and Parro (2014) so that it features PTAs, the costs of non-tariff measures (NTMs), and also covers services trade. We use the gravity equation generated by the model to estimate the Frechet parameters governing the world-wide dispersion of productivities within sectors and the trade cost reductions achieved by existing shallow and deep PTAs. We simulate the model to quantify the trade and welfare effects that a deep TTIP could have. However, the model could also be applied to PTAs, or to trade policy initiatives at the multilateral level.

The main assumption of our quantitative experiment is that the TTIP eliminates tariffs between EU countries and the US and lowers the costs of NTMs by the amount empirically identified for existing bilateral agreements. We find that the potential effects of the TTIP are quite substantial: per capita income could go up by between 2% and 3% in the EU and US, respectively. Average per capita income in the world would also increase, but some non-TTIP countries, mostly in East and Southeast Asia would lose. These effects are driven by a strong amount of trade creation within the TTIP. The framework is ideally suited to study the effects of the agreement on the structure of regional and

global production networks. Our results suggest that the TTIP would strengthen trade in parts and components across the Atlantic, thereby creating a transatlantic production network.

Our framework is close to recent work surveyed by Costinot and Rodríguez-Clare (2014). New quantitative trade theory uses parsimonious trade models and shows how they can be brought to the data by structurally estimating the key parameters. For the time being, these frameworks are only starting to be applied to the analysis of real trade policy initiatives; most existing work still uses more traditional large-scale computable general equilibrium (CGE) models. While the availability of industry-level trade and output data, the information on existing PTAs, and the state-of-the-art of econometric tools has much improved over the last years, there is particular need for further work in at least four areas: First, the unbiased econometric estimation of structural parameters requires quasi-experimental variation and appropriate methods. This remains an important area for future work. Second, top-down approaches to NTMs, as used in this paper, differ from bottom-up assessments based on Francois et al. (2009). Understanding this discrepancy is key if one wants to narrow the interval of welfare predictions found in studies with different approaches to NTMs. Third, both traditional CGE models as well as the frameworks surveyed by Costinot and Rodríguez-Clare (2014) neglect foreign direct investment. This is particularly important in the transatlantic context. Fourth, estimates presented in this paper may underestimate the true effects as they do not account for effects of trade liberalization on the incentives to develop new products or engineer new processes. A tractable dynamic framework would be highly welcome.

References

- Aichele, Rahel and Inga Heiland, "Where is the Value Added? Trade Liberalization and Production Networks," Technical Report, mimeo 2014.
- Alvarez, Fernando and Robert Jr. Lucas, "General equilibrium analysis of the Eaton-Kortum model of international trade," *Journal of Monetary Economics*, September 2007, 54 (6), 1726–1768.
- Anderson, James E., Mario Larch, and Yoto V. Yotov, "On the Effects of the Transatlantic Trade and Investment Partnership on Trade and Capital Accumulation," Technical Report, Boston University 2014.
- Balassa, Bela, "Trade Liberalization and Revealed Comparative Advantage," The Manchester School, 1965, 33 (2), 99–123.
- Baldwin, Richard, "21st Century Regionalism: Filling the gap between 21st century trade and 20th century trade rules," CEPR Policy Insight No. 56, Center for Economic Policy Research May 2011.
- and Dany Jaimovich, "Are Free Trade Agreements contagious?," Journal of International Economics, 2012, 88 (1), 1 – 16.
- Baller, Silja, Trade Effects Of Regional Standards Liberalization : A Heterogeneous Firms Approach, The World Bank, 2007.
- Bhagwati, Jagdish N., *The World Trading System at Risk*, Princeton University Press, 1991.
- Cadot, Olivier, Anne-Célia Disdier, and Lionel Fontagné, "North-South Standards Harmonization and International Trade," CEPR Discussion Papers 8767, Centre for Economic Policy Research January 2012.

- Caliendo, Lorenzo and Fernando Parro, "Estimates of the Trade and Welfare Effects of NAFTA," *Review of Economic Studies*, 2014, *forthcoming*.
- Chen, Maggie Xiaoyang and Aaditya Mattoo, "Regionalism in standards: good or bad for trade?," *Canadian Journal of Economics*, August 2008, 41 (3), 838–863.
- Cipollina, Maria and Luca Salvatici, "Reciprocal Trade Agreements in Gravity Models: A Meta-Analysis," *Review of International Economics*, February 2010, 18 (1), 63–80.
- Costinot, Arnaud and Andrés Rodríguez-Clare, "Trade Theory with Numbers: Quantifying the Consequences of Globalization," in Gita Gopinath, Elhanan Helpman, and Kenneth Rogoff, eds., Handbook of International Economics, Vol. 4 2014, chapter 4, pp. 197–261.
- Dekle, Robert, Jonathan Eaton, and Samuel Kortum, "Global Rebalancing with Gravity: Measuring the Burden of Adjustment," *IMF Staff Papers*, July 2008, 55 (3), 511–540.
- Dür, Andreas, Leonardo Baccini, and Manfred Elsig, "The Design of International Trade Agreements: Introducing a New Database," *Review of International Organizations*, 2014, 9 (3), 353–375.
- Eaton, Jonathan and Samuel Kortum, "Technology, Geography, and Trade," *Econometrica*, September 2002, *70* (5), 1741–1779.
- Egger, Hartmut, Peter Egger, and David Greenaway, "The trade structure effects of endogenous regional trade agreements," *Journal of International Economics*, 2008, 74 (2), 278 – 298.
- Egger, Peter, Joseph Francois, Miriam Manchin, and Douglas Nelson, "Non- Tariff Barriers, Integration, and the Trans-Atlantic Economy," Technical Report, paper prepared for the 60th Panel Meeting of Economic Policy, October 2014, Rom 2014.

- ____, Mario Larch, and Kevin E Staub, "Trade Preferences and Bilateral Trade in Goods and Services: A Structural Approach," CEPR Discussion Papers 9051, C.E.P.R. Discussion Papers July 2012.
- _ , _ , Kevin E. Staub, and Rainer Winkelmann, "The Trade Effects of Endogenous Preferential Trade Agreements," American Economic Journal: Economic Policy, August 2011, 3 (3), 113–43.
- Felbermayr, Gabriel, Benedikt Heid, and Sybille Lehwald, "Transatlantic Free Trade: Winners, Losers, and Necessary Accompanying Measures," Law and Business Review of the Americas, 2013, 19 (4), 445–484.
- _____, ____, Mario Larch, and Erdal Yalcin, "Macroeconomic potentials of transatlantic free trade: A high resolution perspective for Europe and the world," Technical Report, paper prepared for the 60th Panel Meeting of Economic Policy, October 2014 2014.
- Fontagné, Lionel, Julien Gourdon, and Sébastian Jean, "Transatlantic Trade: Whither Partnership, Which Economic Consequences?," CEPII Policy Brief No. 2013-01, CEPII September 2013.
- Francois, Joseph, Hans van Meijl, and Frank van Tongeren, "Trade liberalization in the Doha Development Round," *Economic Policy*, April 2005, 20 (42), 349–391.
- _____, Koen Berden, Saara Tamminen, Martin Thelle, and Paul Wymenga, "NonTariff Measures in EU-US Trade and Investment – An Economic Analysis," Technical Report 2009. Study commissioned by the European Commission, DG Trade, ECORYS Netherland BV.
- _____, Miriam Manchin, Hanna Norberg, Olga Pindyuk, and Patrick Tomberger, "Reducing Transatlantic Barriers to Trade and Investment. An Economic Assessment," Technical Report, Centre for Economic Policy Research 2013.

- Head, Keith and Thierry Mayer, "Gravity Equations: Workhorse, Toolkit, and Cookbook," in G. Gopinath, E. Helpman, and K. Rogoff, eds., *Handbook of International Economics*, Vol. 4 2014, chapter 3, pp. 131–195.
- Hertel, T.W., ed., Global Trade Analysis: Modeling and Applications, Cambridge University Press, 1997.
- Johnson, Robert C. and Guillermo Noguera, "Accounting for intermediates: Production sharing and trade in value added," *Journal of International Economics*, March 2012, 86 (2), 224–236.
- Koopman, Robert, Zhi Wang, and Shang-Jin Wei, "Tracing Value-added and Double Counting in Gross Exports," American Economic Review, 2014, 104 (2), 459–494.
- Krugman, Paul, "Scale Economies, Product Differentiation, and the Pattern of Trade," The American Economic Review, 1980, 70 (5), 950–959.
- Martin, Philippe, Thierry Mayer, and Mathias Thoenig, "The Geography of Conflicts and Regional Trade Agreements," American Economic Journal: Macroeconomics, October 2012, 4 (4), 1–35.
- Wang, Zhi, Shang-Jin Wei, and Kunfu Zhu, "Quantifying International Production Sharing at the Bilateral and Sector Levels," NBER Working Paper No. 19677, National Bureau of Economic Research November 2013.
- WTO, World Trade Report 2012—Trade and public policies: A closer look at non-tariff measures in the 21st century, World Trade Organization, 2012.

Online Appendix to

Going Deep: The Trade and Welfare Effects of TTIP

CESifo Working Paper No. 5150, 2014

Rahel Aichele*, Gabriel Felbermayr † and Inga Heiland ‡

^{*}ifo Institute–Leibniz Institute for Economic Research at the University of Munich, Poschingerstr. 5, 81679 Munich, Germany and CESifo.

[†]ifo Institute–Leibniz Institute for Economic Research at the University of Munich, Ludwig-Maximilians-University Munich, CESifo, and GEP.

[‡]ifo Institute–Leibniz Institute for Economic Research at the University of Munich.

				courto,	Various		1		
	Real	Opennn	$ess \ 1 - \pi_{ii}$	D	m • m	Real inc	come chai	nge	TTTD
	GDP	Initial	Deep	Deep	1arin	Shallow	D: 4	llover	
	p.c.		1 I IP		only		Direct	Indirect	+ Dona
	(III K USD)				(\mathbf{in} (\mathbb{Z})		direct	
Iroland	<u> </u>	34.4	37.1	75	0.0	$\frac{111}{52}$	7.8	7.0	7 /
Lithuania	16.7	39.4	34.4	6.0	0.0	0.2 4 0	6.8	6.0	63
Luxembourg	70.2	54.3	56 1	5.0	-0.0	4.5 5.0	6.1	6.2	5.0
Malta	13.2 21.5	61.9	63 4	0.5 4 0	-0.0	1.0	87	0.2	20.0
Letvie	15.6	$\frac{01.2}{20.7}$	30.8	3.0	0.0	1.2 2.3	0.1 3.7	3.2	20.5
Belgium	10.0 33.7	25.1	37.9	0.0 2.8	-0.0	$\frac{2.5}{1.7}$	3.7 3.7	38	3.2
United States	14 A	8.6	11.3	2.0 2.7	-0.0	1.1 2.1	२.२ २.२	3.0 3.4	0.2 2.8
Cormany	33.0	10.0	21.6	2.1	0.0	$\frac{2.1}{2.0}$	3.0	0.4 2.1	2.0
Austria	35.8	19.4 96.4	21.0	2.0	-0.0	2.0	$\frac{5.0}{2.7}$	2.1	2.0
United Kingdom	33.8	15.1	16.0	2.4 9.3	0.0	1.5 1.7	2.1	2.0	2.0
Bulgaria	10.0	10.1 31 1	32.0	2.3 2.2	0.0	1.7 17	2.0	2.0	2.4 2.6
Slovak Bopublic	10.5	98 1	20.3	2.2	0.0	1.1 9.1	2.6	$\frac{0.5}{2.7}$	2.0
Franco	19.4 31.9	13.0	$\frac{25.3}{15.3}$	2.2	-0.0	2.1	2.0 2.5	2.1	2.5
Portugal	01.2 00.5	19.9	10.0	2.2 2.1	0.0	1.0	2.5	2.0	2.5
Donmark	$\frac{22.0}{24.0}$	10.0	19.9	$\frac{2.1}{2.0}$	0.0	1.4	2.0	2.0 2.7	2.0 9.1
Finland	34.9 22 5	24.9 10.8	20.9	$\frac{2.0}{2.0}$	0.0	1.5	2.0	2.1	2.1
Sweden	35.0 35.0	19.0	21.2 21.0	2.0	0.0	1.0	2.0	2.0	2.2
Notherlands	00.4 97.9	20.3	21.9 20.7	1.0	0.0	1.0	⊿.0 ეე	2.4	2.0
	37.3 96.0	19.4	20.7	1.0	-0.0	1.0	2.2	2.3 2.5	1.9
Deland	20.0 15.5	$\frac{21.2}{10.5}$	22.1	1.0 1.7	0.0	1.4	2.4	2.0	2.0
Fatania	10.0	19.0 20.0	19.9	1.7	0.0	1.0	2.2 2.7	2.0	1.9
Post of North America	19.9	32.2 40.1	32.3 40.0	1.7	0.0	1.0	2.1	2.9 10.1	$\frac{2.0}{1.7}$
Hungary	177	40.1 91.1	40.0	1.0	0.0	1.3	2.1	2.0	1.7
Czoch Popublie	11.1 22.7	01.1 96 9	01.9 06 5	1.0	0.0	0.7	1.9 1.7	2.0	1.0
Spain	20.7	20.2	$\frac{20.3}{17.2}$	1.0	-0.0	0.0	1.7	1.0	1.0
Italy	30.7	10.4 12 4	11.2	1.2 1.2	0.0	0.9	1.7	1.0	1.4
Cummia	00.4 06 5	10.4	14.0 21.5	1.2 1.1	0.0	0.8	1.0	1.7	1.0
Cyprus	20.0 25.6	01.1 20.5	31.0 20.7	1.1	0.0	0.7	2.1	2.3 9.1	1.4
Bomania	20.0	29.0	29.7	1.1	0.0	0.9	2.0 1.6	2.1	1.0
Ronin	11.0	19.9 12.7	20.3	0.9	0.0	0.0	1.0 0.7	1.0	1.1
Toro	1.2	40.7 56 5	40.7 56 5	0.8	0.0	0.5	0.7	5.0 5.7	1.0
Post of Western Africe	0.9	30.3 41.1	30.3 40.0	0.8	0.0	0.4	0.8	5.7 6.0	1.4
Rest of Fastern Furopo		20.8	40.9	0.5	0.0	0.4	0.8	0.9	0.9
Inited Arab Emirated		39.0 36.5	39.8	0.5	0.0	0.4	0.8	0.0 2 1	0.0
Post of South Asia		30.3 96.3	26.4 26.4	0.5	0.0	0.4	0.7	0.1 0.4	1.0
Cuinos	0.0	20.5	20.4	0.4	0.0	0.3	0.4	2.4	1.6
Guinea	0.9	20.0 21.1	20.0	0.4	0.0	0.5	0.5	2.0	1.0
Post of Europo		01.1 02.1	22.0	0.3	0.0	0.2	0.0	2.0	0.7
Coorgia	51	23.1 97.2	23.0 27.3	0.3	0.0	0.5	0.8	2.2	0.4
Ultraino	5.1 7.5	21.5	21.5	0.3	0.0	0.3	0.4 0.7	2.1	0.0
Denomo	10.3	24.9 25.8	24.0 25.0	0.3	0.0	0.5	0.7	$\frac{2.0}{1.7}$	0.5
r anama Croatia	10.0 17 9	20.0 22.0	20.9 99.0	0.0	0.0	0.2	0.5	1.1 0.0	0.1
Sonogal	1 K	20.0 20.⊑	22.9 20 5	0.0	0.0	0.0	0.7	4.4 9.9	0.0
Namibia	1.0	29.0 95 6	29.0 95.6	0.0	0.0	0.2	0.0	2.2 2.7	0.0 9.6
Rotewana	4.9 11 1	20.0 95 /	20.0 25.5	0.0	0.0	0.2	0.4	4.1 1.6	2.0 0.6
DOISWAIIA Doct of FFTA	11.1	20.4 99 E	⊿ə.ə ეэ.э	0.0	0.0	0.1	0.2	1.0	0.0
Rest of Wostown Asia		⊿ə.ə 20.2	∠ə.ə 20.2	0.2	-0.0	0.0	0.8	2.2 2.9	0.7
Rest of Western Asia	11.0	ა∪.ა ვე ი	00.0 20 4	0.2	0.0	0.2	0.4	2.ð 0.1	-0.0
Detarus	11.5	30.2	30.4	0.2	0.0	0.2	0.4	$\angle .1$	0.1

Table A1: Country-specific welfare results, various TTIP scenarios

	Real	Opennn	ess $1 - \pi_{ii}$			Real inc	come char	nge	
	GDP	Initial	Deep	Deep	Tariff	Shallow	Spi	llover	TTIP
	p.c.		TTIP		only		Direct	Indirect	+ Doha
	(in k				(· 07)		direct	
Rost of Control America	(JSD)	20.4	20.4	0.2	0.0	$\frac{111}{0.3}$	1 2	3.0	12.0
Morocco	39	29.4 20.4	29.4 20.3	0.2	-0.0	0.3	1.5	0.9 2.6	12.9 15
Morocco	0.2 0.1	20.4	20.5 38 5	0.2	0.0	0.2	0.0	2.0 2.7	1.0
Ghana	5.4 1.8	$\frac{50.0}{24.7}$	24 6	0.2	0.0	0.5	0.3 0.4	$\frac{2.1}{2.0}$	$1.1 \\ 0.7$
Ostar	107.8	24.1	24.0	0.2	-0.0	0.1	0.4	2.0 2.1	0.1
Norway	50.0	17.9	17.8	0.2	-0.0	0.3	0.0	1.6	0.0
Kenya	13	18.5	18.4	0.2	0.0	0.2 0.2	0.5	1.0 2.1	1.1
Tunisia	1.5 6.4	20.6	29.5	0.2	0.0	0.2 0.2	1.1	2.1	1.1
Ethiopia	0.4	23.0 18.4	25.5 18 /	0.2	0.0	0.2	0.3	13	1.2
Bussia	16.4	10.4	12.4	0.2	-0.0	0.1	0.5	1.5	1.0
Venezuela	10.4 10.1	12.0 12.6	12.0 12.7	0.2	-0.0	0.1	0.5	1.0	1.0
Kyrgyzstan	2.2	20.0	20.8	0.2	0.0	0.1	0.3	1.0	0.5
Hong Kong	$\frac{2.2}{37.0}$	25.5 26.5	26.5	0.2	0.0	0.1	0.2	2.5	0.5
South Africa	77	20.0 14 3	14.9	0.2	-0.0	0.1	0.4	2.5	0.0
Fount	1.1	14.0 10.7	14.2 10.7	0.2	-0.0	0.1	0.4	1.0	3.0
Albania	4.5 6.1	19.1 98.1	19.7 28.1	0.2	-0.0	0.1	0.4	1.3 1 7	0.1
Caribboan	0.1	20.1 16.3	20.1 16.2	0.1	0.0	0.2	0.5	1.7	0.1
Armonia	5.4	10.0	10.2	0.1	-0.0	0.1	0.5	1.4	0.4
Paraguay	3.4	19.9 22.0	19.8	0.1	0.0	0.1	0.2	1.0	1.6
Tanaguay	1.0	22.9 93.1	22.9 23.1	0.1	-0.0	0.0	0.1	2.0	1.0
Bahrain	20.7	20.1	20.1	0.1	-0.0	0.1	0.4	2.8	1.5
Nepal	23.1	17.1	17.1	0.1	-0.0	0.2	0.4	2.0	1.5
Canada	35.7	16.5	15.8	0.1	-0.0	0.1	0.1	1.1	1.5
Israal	24 A	10.0 22.5	10.0 22.3	0.1	-0.0	0.1	0.5	1.0	0.2
Bwanda	24.4	18.6	18.7	0.1	0.0	0.2	0.4	1.0	0.4
Laos	2.1	21.6	21.6	0.1	0.0	0.1	0.2	1.2 2.4	0.8
Best of North Africa	2.1	16.4	16 A	0.1	-0.0	0.1	0.2	2.4 1.6	-0.2
Kazakhstan	11.8	10.4 17.8	10.4 17.8	0.1	-0.0	0.1	0.5	1.0	-0.2
Nigoria	1 0	25.0	25.0	0.1	-0.0	0.1	0.5	1.0	0.2
Madagascar	1.9	$\frac{25.0}{18.1}$	17.9	0.1	0.0	0.1	0.2	1.4	-0.2
Turkey	127	15.6	15.5	0.1	-0.0	0.1	0.0	1.5	0.2
Switzerland	12.1 49.4	25.1	24.8	0.1	-0.0	0.1	0.4	1.2	0.5
Cameroon	18	12.6	12.5	0.1	0.0	0.2	0.0	1.0	0.0
Burkina Faso	0.9	12.0 17.6	12.0 177	0.1	0.0	0.1	0.4	1.0	0.4
Colombia	7.3	10.7	10.7	0.1	-0.0	0.1	0.2	1.2	0.0
Honduras	2.0	20.7	20.1	0.1	-0.0	0.1	0.2	$25^{1.1}$	-0.0
Rest of Eastern Δ frice	2.5	25.1 15.0	1/ 9	0.1	-0.0	0.1	0.0	$\frac{2.5}{1.7}$	-0.0
Central Africa		20.4	20.3	0.1	0.0	0.1	0.2 0.4	1.7	13
Australia	36.5	10.4	10.6	0.1	-0.0	0.2	0.4 0.2	1.1	0.5
El Salvador	1 1	22.3	21.0	0.1	-0.0	0.1	0.2	1.1	0.0
Uganda	1.1	17.2	17.2	0.1	0.0	0.0	0.1	1.0	0.7
New Zealand	27.2	12.7	12.6	0.1	-0.0	0.0	0.2 0.3	1.5	0.1
India	2.8	13.1	13.0	0.1	-0.0	0.1	0.0	0.0	1.5
Pakistan	$\frac{2.0}{2.3}$	15.5	15.3	0.1	-0.0	0.1	0.2	1.2	0.4
Rest of Oceania	2.0	20.6	20.4	0.1	-0.0	0.1	0.2	19	-0.0
Malawi	0.7	$\frac{20.0}{20.6}$	20.4	0.1	0.0	0.1	0.0	2.3	7.3
Rest of the World	0.1	14.6	14.3	0.1	0.0	0.1	0.1	1.3 1.2	0.0
LUCOU OF UNIO FROMU		· · ·	± ±.0	U.1	0.0	0.1	0.0	±.4	0.0

Table A1: Country-specific welfare results, various TTIP scenarios

	Real	Opennn	ess $1 - \pi_{ii}$			Real inco	ome chang	ge	
	GDP	Initial	Deep	Deep	Tariff	Shallow	Spill	over	TTIP
	p.c.		TTIP	-	only		Direct	In-	+ Doha
	(in k							direct	
	ÚSD)				(i:	n %)			
Bolivia	3.6	15.7	15.7	0.0	0.0	0.0	0.2	1.0	0.4
Guatemala	3.9	20.1	20.0	0.0	-0.0	0.1	0.4	1.7	-0.1
Zimbabwe	3.7	36.6	36.5	0.0	0.0	0.1	0.4	3.6	5.3
Ecuador	6.2	17.1	17.2	0.0	0.0	0.0	0.4	1.4	0.1
Oman	23.4	32.5	32.4	0.0	-0.0	0.2	0.4	2.9	0.5
Singapore	51.9	43.2	42.8	0.0	-0.0	0.3	0.7	3.5	0.7
Uruguay	9.6	17.8	17.8	0.0	0.0	0.0	0.4	1.5	1.8
Iran	12.2	17.0	17.0	-0.0	-0.0	0.1	0.2	1.4	0.1
Brazil	8.1	7.0	6.9	-0.0	-0.0	0.0	0.1	0.6	0.1
Cote d'Ivoire	1.4	19.7	19.6	-0.0	-0.0	0.0	0.7	2.0	0.1
Sri Lanka	3.7	23.1	22.8	-0.0	-0.0	0.1	0.3	1.6	1.2
Argentina	11.6	12.0	11.9	-0.0	-0.0	0.0	0.3	1.3	0.2
Vietnam	2.6	41.4	41.2	-0.1	-0.0	0.2	0.7	3.9	0.2
Rest of Frm. Sov. Union		18.9	18.9	-0.1	-0.0	-0.0	0.2	2.2	-0.6
Mexico	12.9	15.2	14.7	-0.1	-0.0	-0.0	0.3	1.1	-0.1
South Central Africa		28.1	28.0	-0.1	0.0	0.1	0.4	2.2	1.0
Indonesia	3.5	13.4	13.3	-0.1	-0.0	-0.0	0.1	0.9	0.6
Rest of East Asia		20.7	20.6	-0.1	-0.0	-0.0	0.2	1.0	-0.1
Rest of Southeast Asia		17.0	17.0	-0.1	-0.0	-0.0	0.1	1.3	-0.0
Zambia	1.7	15.6	15.5	-0.1	-0.0	-0.1	0.1	1.2	0.5
Saudi Arabia	21.5	27.9	27.8	-0.1	-0.0	0.1	0.3	1.9	0.3
Japan	31.7	8.3	8.1	-0.1	-0.0	-0.1	-0.0	0.4	0.3
Bangladesh	1.3	13.9	13.6	-0.1	-0.0	0.1	0.3	1.4	-0.6
Peru	7.0	10.3	10.1	-0.1	-0.0	-0.1	0.1	0.9	-0.3
Rest of South America		25.0	24.5	-0.2	-0.0	-0.2	0.6	2.4	7.0
Kuwait	62.9	24.0	24.0	-0.2	-0.0	0.1	0.3	2.0	0.9
Mozambique	0.7	31.0	30.9	-0.2	-0.0	-0.2	0.4	2.4	1.5
Chile	13.7	17.3	17.1	-0.2	-0.0	-0.1	0.3	1.5	-0.4
Costa Rica	9.3	26.2	25.8	-0.2	-0.0	0.1	0.9	2.6	-0.4
Korea, Republic of	26.0	17.7	17.4	-0.2	-0.0	-0.1	0.1	1.0	0.6
Philippines	3.2	24.6	24.2	-0.2	-0.0	0.0	0.2	1.7	-0.2
China	6.6	10.8	10.6	-0.3	-0.0	0.0	0.2	1.1	0.4
Mongolia	4.4	32.0	31.9	-0.3	-0.0	-0.1	0.2	2.4	0.4
Thailand	8.0	28.9	28.6	-0.3	-0.0	-0.0	0.2	2.2	0.3
Taiwan	28.2	26.9	26.6	-0.4	-0.0	-0.0	0.2	1.6	1.0
Malaysia	12.2	30.3	29.8	-0.5	-0.0	-0.0	0.4	3.5	0.6
Rest of $SACU^a$		19.4	19.0	-0.7	-0.0	-0.0	-0.0	1.1	5.3
Cambodia	1.8	34.2	33.2	-0.9	-0.0	0.2	0.4	3.1	-1.0

Table A1: Country-specific welfare results, various TTIP scenarios

Note: The table provides an overview of the country-specific effects of different TTIP scenarios. a SACU stands for Southern African Customs Union.

A Regional and sectoral breakdown

Countries and regions in GTAP The 114 countries in GTAP are: Albania, Argentina, Armenia, Australia, Austria, Azerbaijan, Bangladesh, Belarus, Belgium, Benin, Bolivia, Botswana, Brazil, Bulgaria, Burkina Faso, Cambodia, Cameroon, Canada, Chile, China, Colombia, Costa Rica, Cote d Ivoire, Croatia, Cyprus, Czech Republic, Denmark, Ecuador, Egypt, El Salvador, Estonia, Ethiopia, Finland, France, Georgia, Germany, Ghana, Greece, Guatemala, Guinea, Honduras, Hong Kong, Hungary, India, Indonesia, Iran, Ireland, Israel, Italy, Japan, Kazakhstan, Kenya, Kuwait, Kyrgyzstan, Laos, Latvia, Lithuania, Luxembourg, Madagascar, Malawi, Malaysia, Malta, Mauritius, Mexico, Mongolia, Morocco, Mozambique, Namibia, Nepal, Netherlands, New Zealand, Nicaragua, Nigeria, Norway, Oman, Pakistan, Panama, Paraguay, Peru, Philippines, Poland, Portugal, Qatar, Russian Federation, Rwanda, Saudi Arabia, Senegal, Singapore, Slovakia, Slovenia, South Africa, South Korea, Spain, Sri Lanka, Sweden, Switzerland, Taiwan, Tanzania, Thailand, Togo, Tunisia, Turkey, Uganda, Ukraine, United Arab Emirates, United Kingdom, United States of America, Uruguay, Venezuela, Vietnam, Zambia and Zimbabwe.

Other countries are aggregated into 20 regional composites such as, e.g., Rest of Southeast Asia comprising Brunei Darussalam, Myanmar, and Timor Leste.

Region listing The following list provides an overview of the regional breakdown we use to present the results. Note that the simulations are performed with the 134 original GTAP countries and regions, though.

 ASEAN: Indonesia, Cambodia, Laos, Malaysia, Philippines, Singapore, Thailand, Vietnam, Rest of Southeast Asia (Brunei Darussalam, Myanmar, Timor Leste¹)

¹Timor Leste is not part of ASEAN. However, in GTAP's regional breakdown it is included in the region "Rest of Southeast Asia" together with Brunei Darussalam and Myanmar, both ASEAN member states. Therefore, we assigned the region "Rest of Southeast Asia" to ASEAN.

- Central Asia: Azerbaijan, Georgia, Kyrgyzstan, Ukraine, Mongolia, Rest of Eastern Europe (Moldova), Rest of Former Soviet Union (Tajikistan, Turkmenistan, Uzbekistan)
- China: China, Hong Kong
- East Asia: Japan, South Korea, Taiwan
- EFTA: Norway, Switzerland, Rest of EFTA (Iceland, Liechtenstein)
- Eurasian Customs Union: Armenia, Belarus, Kazakhstan, Russian Federation
- Latin America & Caribbean: Bolivia, Ecuador, Guatemala, Honduras, Nicaragua, Panama, El Salvador, Rest of Central America (Belize), Rest of South America (Falkland Islands, French Guiana, Guyana, South Georgia and the South Sandwich Islands, Suriname), Caribbean (Anguilla, Antigua & Barbuda, Aruba, Bahamas, Barbados, Cayman Islands, Cuba, Dominica, Dominican Republic, Grenada, Haiti, Jamaica, Montserrat, Netherlands Antilles, Puerto Rico, Saint Kitts and Nevis, Saint Lucia, Saint Vincent and the Grenadines, Trinidad and Tobago, Turks and Caicos Islands, Virgin Islands British, Virgin Islands U.S.)
- MERCOSUR: Argentina, Brazil, Paraguay, Uruguay, Venezuela
- Middle East & North Africa: Israel, Egypt, Iran, Morocco, Tunisia, Rest of North Africa (Algeria, Libyan Arab Jamahiriya, Western Sahara), Rest of Western Asia (Iraq, Jordan, Lebanon, Palestinian Territory Occupied, Syrian Arab Republic, Yemen)
- Oil exporters: United Arab Emirates, Bahrain, Kuwait, Oman, Qatar, Saudi Arabia
- Pacific Alliance: Chile, Colombia, Costa Rica, Mexico, Peru

- Rest of Europe: Albania, Croatia, Rest of Europe (Andorra, Bosnia and Herzegovina, Faroe Islands, Gibraltar, Guernsey, Holy See, Isle of Man, Jersey, Former Yugoslav Republic of Macedonia, Monaco, Montenegro, San Marino, Serbia)
- Rest of Oceania: Rest of Oceania (American Samoa, Cook Islands, Fiji, French Polynesia, Guam, Kiribati, Marshall Islands, Micronesia Federated States of, Nauru, New Caledonia, Niue, Northern Mariana Islands, Palau, Papua New Guinea, Pitcairn, Samoa, Solomon Islands, Tokelau, Tonga, Tuvalu, United States Minor Outlying Islands, Vanuatu, Wallis and Futuna)
- Rest of World: Rest of East Asia (Democratic People's Republic of Korea, Macao), Rest of South Asia (Afghanistan, Bhutan, Maldives), Rest of North America (Bermuda, Greenland, Saint Pierre and Miquelon), Rest of the World (Antarctica, Bouvet Island, British Indian Ocean Territory, French Southern Territories)
- South Asia: Bangladesh, India, Sri Lanka, Nepal, Pakistan
- Southern African Customs Union: Botswana, Namibia, South Africa, Rest of South African Customs Union (Lesotho, Swaziland)
- Sub-Saharan Africa: Benin, Burkina Faso, Ivory Coast, Ethiopia, Cameroon, Ghana, Guinea Bissau, Kenya, Madagascar, Mozambique, Nigeria, Malawi, Mauritius, Rwanda, Senegal, Togo, Tanzania, Uganda, Rest of Western Africa (Cape Verde, Gambia, Liberia, Mali, Mauritania, Niger, Saint Helena, Ascension and Tristan Da Cunha, Sierra Leone), Central Africa (Central African Republic, Chad, Congo, Equatorial Guinea, Gabon, Sao Tome and Principe), South Central Africa (Angola, Congo the Democratic Republic of), Rest of Eastern Africa (Burundi, Comoros, Djibouti, Eritrea, Mayotte, Seychelles, Somalia, Sudan)

	1 able A2: (Jverview of sectors
	GTAP sector	Sector description
1	1-14,19,20,23	Agriculture & Food
2	15-18	Mining
3	21,22,24-26	Food, processed
4	27 + 28	Textiles & Wearing apparel
5	29	Leather
6	30	Wood
7	31	Paper
8	32	Petroleum
9	33	Chemicals
10	34	Mineral products
11	35	Ferrous metals
12	36	Metals nec
13	37	Metal products
14	38 + 39	Transport equipment
15	40	Electronics
16	41	Machinery nec
17	42	Manufactures nec
18	43	Electricity
19	44	Gas
20	45	Water
21	46	Construction
22	47	Trade services
23	48	Transport nec
24	49	Water Transport
25	50	Air Transport
26	51	Communication
27	52	Financial services nec
28	53	Insurance
29	54	Business services nec
30	55	Recreational services
31	56	Other services
32	57	Dwellings

Table A2: Overview of sectors

Note: The table provides an overview of the sectoral aggregation used throughout this publication. There are one agricultural and one resource extraction sector, 15 manufacturing sectors and 15 services sectors.

B Detailed estimation results

	(1)	(2)	(3)	(4)	(5)	(0)	(2)	(8)
	Agriculture	Mining	Food	Textiles	Leather	Wood	Paper	Petroleum
n tariff	-2.861***	-20.066***	-3.140***	-4.342***	-5.046^{***}	-5.684***	-4.848***	-9.538***
	(0.636)	(2.791)	(0.605)	(0.860)	(0.920)	(0.806)	(1.012)	(3.335)
hallow PTA	1.442^{***}	-0.975	-0.118	-0.321	-0.204	1.137^{**}	0.826	-1.702
	(0.458)	(0.793)	(0.532)	(0.533)	(0.650)	(0.536)	(0.571)	(1.116)
eep PTA	0.689	4.292^{***}	1.564^{**}	2.357^{***}	2.049^{**}	0.473	0.021	3.527^{**}
	(0.630)	(1.087)	(0.712)	(0.700)	(0.831)	(0.707)	(0.762)	(1.432)
n distance	-0.781***	-1.129^{***}	-1.277***	-1.188^{***}	-1.067^{***}	-1.319^{***}	-1.836^{***}	-2.216^{***}
	(0.079)	(0.128)	(0.070)	(0.067)	(0.081)	(0.071)	(0.074)	(0.143)
ontiguity	1.248^{***}	1.088^{***}	0.779^{***}	0.956^{***}	1.114^{***}	0.799^{***}	0.370^{**}	0.954^{***}
	(0.164)	(0.234)	(0.160)	(0.167)	(0.185)	(0.164)	(0.181)	(0.261)
ommon language	0.496^{***}	0.230	0.697^{***}	0.780^{***}	0.748^{***}	0.605^{***}	1.150^{***}	0.020
	(0.100)	(0.161)	(0.093)	(0.089)	(0.103)	(0.093)	(0.100)	(0.217)
ver colonial rel.	1.094^{***}	1.519^{***}	1.094^{***}	0.910^{***}	1.010^{***}	0.918^{***}	0.787^{***}	0.934^{***}
	(0.147)	(0.234)	(0.155)	(0.157)	(0.176)	(0.138)	(0.159)	(0.280)
urrent colonial rel.	0.196	1.644^{***}	-1.708	0.206	0.552	-0.300	-3.016	1.436
	(0.271)	(0.497)	(2.179)	(1.923)	(2.177)	(2.408)	(2.414)	(1.329)
ver same country	0.310	0.512	0.669^{***}	0.284	0.823^{***}	0.404^{*}	0.535^{**}	0.555
	(0.226)	(0.354)	(0.219)	(0.270)	(0.288)	(0.240)	(0.248)	(0.393)
bservations	7938	5228	8032	8547	6738	7336	7684	4101
5	0.607	0.502	0.663	0.744	0.692	0.711	0.732	0.493
MSE	2.238	3.004	2.080	1.965	2.123	2.031	2.146	3.071

Note: The table shows gravity estimates for non services sectors with importer and exporter dummies. Standard errors (in
parantheses) are heteroskedasticity-robust. PTA dummies instrumented with single contagion index a la Martin et al. (2012)
or Baldwin and Jaimovich (2012), historical and recent war frequency and lagged average variables for political similarity
(average of 2000-2005). *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient
on the coefficient on the PTA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified
with data on the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches
to one for depth $>= 0$, and the deep PTA dummy switches to one for depth $>= 4$. I.e., the shallow dummy gives the effect of
having an PTA, the deep dummy gives the additional effect of having a deep agreement.

		Table A	4: IV gravity e	stimates ma	anufacturing	sectors, cont'	d.		
	(9)	(10) $Minorele$	(11) Formalis motals	(12) M_{otol} n_{oo}	(13) Motel mod	(14)	(15) $\mathbb{P}_{100tronioc}$	(16) $M_{achinomy noo}$	(17) Manufactunes noo
	CHEIIICAIS	INTITIC ALS	rerrous metals	INTERAL TIEC	merai prou.	rrausp. equip	Electrollics	maculitiery nec	Mallulactures nec
Ln tariff	-11.210^{***}	-8.872***	-10.456^{***}	-10.756^{***}	-5.725^{***}	-1.484^{*}	-4.321^{***}	-7.186^{***}	-3.867^{***}
	(1.097)	(0.890)	(1.355)	(1.901)	(1.170)	(0.796)	(1.498)	(1.022)	(0.747)
Shallow PTA	0.194	0.430	-0.137	1.706^{***}	0.059	2.210^{***}	-0.921	0.328	0.526
	(0.448)	(0.533)	(0.636)	(0.643)	(0.559)	(0.608)	(0.617)	(0.483)	(0.571)
Deep PTA	1.055^{*}	0.287	2.245^{***}	-0.831	0.859	-1.175	2.032^{**}	0.550	-0.069
	(0.622)	(0.692)	(0.845)	(0.818)	(0.754)	(0.768)	(0.829)	(0.658)	(0.757)
Ln distance	-1.437^{***}	-1.592^{***}	-1.398^{***}	-1.525^{***}	-1.626^{***}	-1.226^{***}	-1.213^{***}	-1.373^{***}	-1.262^{***}
	(0.064)	(0.073)	(0.084)	(0.098)	(0.066)	(0.080)	(0.071)	(0.063)	(0.067)
Contiguity	0.792^{***}	0.986^{***}	0.927^{***}	0.169	0.483^{***}	0.757^{***}	0.515^{***}	0.618^{***}	0.419^{**}
	(0.164)	(0.174)	(0.169)	(0.199)	(0.164)	(0.175)	(0.190)	(0.169)	(0.163)
Common language	0.787^{***}	0.728^{***}	0.389^{***}	0.487^{***}	0.853^{***}	0.579^{***}	0.616^{***}	0.749^{***}	0.856^{***}
	(0.082)	(0.096)	(0.123)	(0.131)	(0.092)	(0.105)	(0.094)	(0.078)	(0.087)
Ever colonial rel.	0.456^{***}	0.768^{***}	0.742^{***}	1.100^{***}	0.926^{***}	0.632^{***}	1.135^{***}	0.813^{***}	0.840^{***}
	(0.127)	(0.149)	(0.176)	(0.171)	(0.149)	(0.165)	(0.177)	(0.137)	(0.158)
Current colonial rel.	-0.276	-2.103	1.140	1.531	-0.903	0.006	-0.483	0.334	0.536
	(2.174)	(3.625)	(1.700)	(1.060)	(2.191)	(1.172)	(3.096)	(2.344)	(2.245)
Ever same country	0.377	0.497^{*}	0.564^{**}	-0.180	0.423^{*}	0.382	0.116	0.357	0.569^{**}
	(0.231)	(0.261)	(0.267)	(0.293)	(0.245)	(0.275)	(0.316)	(0.242)	(0.236)
Observations	9027	7094	5816	5498	7750	7336	7944	9091	7811
$ m R^2$	0.767	0.732	0.659	0.641	0.746	0.707	0.757	0.808	0.742
RMSE	1.953	1.957	2.272	2.345	1.998	2.263	2.072	1.828	1.900
Note: The table shows	oravity estim	ates for non s	ervices sectors wit	th importer a	nd exnorter du	mmies Standard	l errors (in nar	antheses) are hete	roskedasticity.

The coefficient on the coefficient on the PTA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified with data on dummy switches to one for depth >= 4. I.e., the shallow dummy gives the effect of having an PTA, the deep dummy gives the additional effect of having a the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth >= 0, and the deep PTA robust. PTA dummies instrumented with single contagion index a la Martin et al. (2012) or Baldwin and Jaimovich (2012), historical and recent war frequency and lagged average variables for political similarity (average of 2000-2005). *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. deep agreement.

	T	able A5: I	V gravity	estimates ser	vice secto	rs	
	(18) Electricity	(19) Gas	(20) Water	(21) Construction	(22)Trade	(23) Transport nec	(24) Water transport
Shallow PTA	0.137^{***}	0.025	0.086^{***}	0.024	0.036	0.060*	-0.028
	(0.048)	(0.056)	(0.030)	(0.061)	(0.038)	(0.031)	(0.056)
Deep PTA	0.192^{**}	0.354^{***}	0.250^{***}	0.304^{***}	0.452^{***}	0.267^{***}	0.479^{***}
	(0.082)	(0.092)	(0.055)	(0.092)	(0.061)	(0.054)	(0.087)
Ln distance	-0.041	0.051^{*}	0.024	-0.031	-0.003	-0.008	0.061^{**}
	(0.030)	(0.030)	(0.017)	(0.026)	(0.018)	(0.016)	(0.024)
Contiguity	2.070^{***}	0.525^{***}	0.507^{***}	0.321^{***}	0.426^{***}	0.413^{***}	0.468^{***}
	(0.157)	(0.086)	(0.065)	(0.081)	(0.063)	(0.058)	(0.069)
Common language	-0.031	0.010	-0.038^{*}	0.044	-0.057**	-0.073^{***}	-0.037
	(0.035)	(0.035)	(0.020)	(0.035)	(0.024)	(0.020)	(0.032)
Ever colonial rel.	0.389^{***}	0.334^{***}	0.366^{***}	0.190^{***}	0.405^{***}	0.387^{***}	0.256^{***}
	(0.088)	(0.066)	(0.059)	(0.071)	(0.056)	(0.054)	(0.070)
Current colonial rel.	0.625	1.536^{*}	0.690	-0.108	0.900	0.713	0.320
	(0.940)	(0.836)	(0.928)	(1.185)	(1.018)	(0.870)	(1.087)
Ever same country	0.188	0.425^{***}	0.193^{*}	0.062	0.123	0.159^{*}	0.175
	(0.233)	(0.158)	(0.111)	(0.126)	(0.105)	(0.093)	(0.118)
Observations	11918	9161	12571	12656	12656	12656	12656
${ m R}^2$	0.971	0.963	0.964	0.905	0.958	0.959	0.917
RMSE	0.748	0.831	0.557	0.993	0.640	0.553	0.896

(in parantheses) are	eı, respectively. 1 ne ces a la Martin et al.	r et al. (2014) , which	for depth>= 4. I.e.,	r 32 "Dwellings" not	
s. Standard errors	the 10, 5 and 1% lev with contagion inde:	oth of PTAs from Dü	imy switches to one	ep agreement. Secto	
exporter fixed effect	stical significance at mmies instrumented	with data on the der	the deep PTA dur	effect of having a de	
vith importer and e	a for δ/θ . PTA dur	p PTAs is classified	for depth ≥ 0 , and	ives the additional of	
ices sectors with w	pair level. ', and parameter estimat	or shallow and deer	y switches to one f	the deep dummy g	flows reported.
estimates for serv	rea at the country- PTA dummy is the	(12). The dummy f	hallow PTA dumm	of having an PTA,	there are no trade
shows IV gravity	-robust and cluster coefficient on the 1	and Jaimovich (20)	7 and where the s	ny gives the effect a	timations because
Note: The table	neveroskedasucuty coefficient on the	(2012) or Baldwin	ranges from 0 to	the shallow dumn	included in the est

	120		í .	1007	(00)	~~~~	(10)
	(25) Air transport	(26) Communication	(27) Financial serv. nec	(28) Insurance	(29) Business serv.	(30) Recreational serv.	(31) Other
Shallow PTA	0.064	0.032	0.048	0.038	0.066^{*}	0.030	0.086^{**}
	(0.039)	(0.027)	(0.039)	(0.031)	(0.036)	(0.033)	(0.035)
Deep PTA	-0.025	0.076	0.303^{***}	0.136^{***}	0.322^{***}	0.192^{***}	0.268^{***}
ĸ	(0.059)	(0.046)	(0.064)	(0.052)	(0.056)	(0.054)	(0.059)
Ln distance	-0.000	-0.007	-0.040^{**}	0.006	-0.013	-0.025	-0.016
	(0.017)	(0.016)	(0.019)	(0.016)	(0.017)	(0.017)	(0.018)
Contiguity	0.337^{***}	0.386^{***}	0.297^{***}	0.391^{***}	0.325^{***}	0.398^{***}	0.371^{***}
)	(0.053)	(0.058)	(0.068)	(0.058)	(0.061)	(0.060)	(0.064)
Common language	-0.034	-0.024	-0.061^{**}	-0.038*	-0.044**	-0.029	-0.054^{**}
	(0.021)	(0.018)	(0.025)	(0.020)	(0.022)	(0.021)	(0.023)
Ever colonial rel.	0.212^{***}	0.237^{***}	0.279^{***}	0.178^{***}	0.281^{***}	0.272^{***}	0.288^{***}
	(0.041)	(0.050)	(0.070)	(0.056)	(0.050)	(0.053)	(0.060)
Current colonial rel.	1.325	1.045	0.698	1.629^{**}	0.833	1.566^{***}	0.641
	(1.111)	(1.014)	(0.893)	(0.697)	(1.178)	(0.589)	(0.995)
Ever same country	-0.001	0.089	0.200^{*}	0.223^{**}	0.105	0.106	0.282^{***}
	(0.079)	(0.091)	(0.117)	(0.100)	(0.090)	(0.099)	(0.105)
Observations	12656	12656	12656	12656	12656	12656	12656
$ m R^2$	0.956	0.966	0.954	0.963	0.965	0.961	0.941
RMSE	0.606	0.484	0.674	0.550	0.600	0.586	0.613

heteroskedasticity-robust and clustered at the country-pair level. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the coefficient on the PTA dummy is the parameter estimate for δ/θ . PTA dummies instrumented with contagion indexes a la Martin et al. (2012) or Baldwin and Jaimovich (2012). The dummy for shallow and deep PTAs is classified with data on the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>= 0, and the deep PTA dummy switches to one for depth>= 4. I.e., Note: The table shows IV gravity estimates for services sectors with with importer and exporter fixed effects. Standard errors (in parantheses) are the shallow dummy gives the effect of having an PTA, the deep dummy gives the additional effect of having a deep agreement. Sector 32 "Dwellings" not included in the estimations because there are no trade flows reported.

	TAUUT	a GUU : IA	TUGD GULVDT	TITICONDITI	ant monorni	STOUDAS		
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	Agriculture	Mining	Food	Textiles	Leather	Wood	Paper	$\operatorname{Petroleum}$
In tariff	-3.267^{***}	-16.429^{***}	-3.252^{***}	-4.667***	-5.367^{***}	-6.609***	-5.157^{***}	-7.165^{**}
	(0.608)	(2.197)	(0.500)	(0.608)	(0.793)	(0.780)	(0.938)	(2.833)
Shallow PTA	0.229^{***}	-0.015	0.155^{**}	0.105	0.121	0.093	0.107	-0.058
	(0.078)	(0.133)	(0.076)	(0.070)	(0.085)	(0.076)	(0.082)	(0.167)
Deep PTA	0.731^{***}	0.648^{***}	0.185^{*}	0.591^{***}	0.706^{***}	0.479^{***}	0.366^{***}	0.173
	(0.104)	(0.173)	(0.098)	(0.091)	(0.113)	(0.099)	(0.104)	(0.202)
In distance	-1.118^{***}	-1.614^{***}	-1.488***	-1.422^{***}	-1.239^{***}	-1.594^{***}	-1.960^{***}	-2.471^{***}
	(0.050)	(0.083)	(0.046)	(0.045)	(0.055)	(0.049)	(0.051)	(0.099)
Contiguity	1.166^{***}	0.629^{***}	0.597^{***}	0.737^{***}	0.957^{***}	0.711^{***}	0.375^{**}	0.587^{**}
	(0.148)	(0.206)	(0.147)	(0.155)	(0.178)	(0.157)	(0.166)	(0.242)
Jommon language	0.520^{***}	0.182	0.673^{***}	0.712^{***}	0.672^{***}	0.570^{***}	1.131^{***}	-0.229
	(0.095)	(0.150)	(0.090)	(0.084)	(0.100)	(0.088)	(0.098)	(0.184)
Ever colonial rel.	1.117^{***}	1.187^{***}	0.978^{***}	0.767^{***}	0.910^{***}	0.965^{***}	0.841^{***}	0.743^{***}
	(0.139)	(0.204)	(0.140)	(0.142)	(0.156)	(0.131)	(0.140)	(0.266)
Jurrent colonial rel.	0.138	1.598^{**}	-1.790	0.280	0.642	-0.249	-2.994	2.132
	(0.286)	(0.620)	(2.368)	(2.307)	(2.428)	(2.415)	(2.408)	(1.381)
Ever same country	0.170	0.523	0.691^{***}	0.370'	0.743^{***}	0.293	0.466^{**}	0.509
	(0.225)	(0.357)	(0.213)	(0.251)	(0.286)	(0.232)	(0.228)	(0.388)
Observations	8132	5384	8225	8766	6948	7534	7874	4191
ξ ²	0.622	0.541	0.670	0.756	0.700	0.717	0.736	0.529
P-stat	66.232	36.234	76.911	120.693	81.484	94.386	110.141	
RMSE	2.229	2.940	2.082	1.946	2.133	2.030	2.157	3.052

robust. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the coefficient on the PTA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified with data on the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth >= 0, and the deep PTA dummy switches to one for depth >= 4. I.e., the shallow dummy Note: The table shows gravity estimates for non services sectors with importer and exporter dummies. Standard errors (in parantheses) are heteroskedasticitygives the effect of having an PTA, the deep dummy gives the additional effect of having a deep agreement.

		Table A8	: OLS gravity (estimates n	lanufacturing	g sectors, cont	'n.		
	(9) Chemicals	(10) Minerals	(11) Ferrous metals	(12) Metal nec	(13) Metal prod.	(14) Transp. equip	(15) Electronics	(16) Machinery nec	(17) Manufactures nec
Ln tariff	-10.786^{***}	-9.118^{***}	-10.092^{***}	-12.575***	-5.450^{***}	-2.377***	-2.587**	-6.952***	-3.962^{***}
	(0.861)	(0.857)	(1.269)	(1.517)	(0.922)	(0.727)	(1.173)	(0.860)	(0.725)
Shallow PTA	0.203^{***}	0.088	0.247^{**}	0.330^{***}	0.183^{**}	0.316^{***}	0.208^{***}	0.174^{***}	0.343^{***}
	(0.065)	(0.081)	(0.100)	(0.106)	(0.074)	(0.086)	(0.075)	(0.064)	(0.070)
Deep PTA	0.430^{***}	0.247^{**}	0.671^{***}	0.370^{***}	0.295^{***}	0.475^{***}	0.332^{***}	0.257^{***}	0.165^{*}
	(0.087)	(0.101)	(0.127)	(0.133)	(0.095)	(0.108)	(0.102)	(0.082)	(060.0)
Ln distance	-1.553^{***}	-1.675^{***}	-1.605^{***}	-1.647^{***}	-1.692^{***}	-1.425^{***}	-1.246^{***}	-1.466^{***}	-1.262^{***}
	(0.042)	(0.049)	(0.058)	(0.065)	(0.046)	(0.051)	(0.046)	(0.040)	(0.045)
Contiguity	0.709^{***}	0.979^{***}	0.760^{***}	0.226	0.429^{***}	0.825^{***}	0.353^{**}	0.580^{***}	0.450^{***}
	(0.152)	(0.164)	(0.157)	(0.180)	(0.153)	(0.155)	(0.174)	(0.157)	(0.152)
Common language	0.760^{***}	0.708^{***}	0.350^{***}	0.569^{***}	0.824^{***}	0.630^{***}	0.558^{***}	0.719^{***}	0.816^{***}
	(0.080)	(0.093)	(0.115)	(0.118)	(0.089)	(0.099)	(0.089)	(0.076)	(0.085)
Ever colonial rel.	0.415^{***}	0.780^{***}	0.590^{***}	1.205^{***}	0.891^{***}	0.829^{***}	0.977^{***}	0.803^{***}	0.885^{***}
	(0.113)	(0.139)	(0.154)	(0.161)	(0.130)	(0.146)	(0.144)	(0.125)	(0.143)
Current colonial rel.	-0.255	-2.069	1.051	1.371	-0.879	0.042	-0.474	0.365	0.528
	(2.322)	(3.688)	(2.002)	(0.981)	(2.328)	(0.941)	(3.459)	(2.433)	(2.218)
Ever same country	0.385^{*}	0.461^{*}	0.518^{**}	-0.340	0.428^{*}	0.179	0.238	0.366	0.516^{**}
	(0.224)	(0.259)	(0.257)	(0.281)	(0.236)	(0.245)	(0.288)	(0.228)	(0.219)
Observations	9248	7278	6001	5682	7943	7510	8162	9316	8025
$ m R^2$	0.767	0.731	0.666	0.653	0.747	0.726	0.770	0.808	0.744
F-stat	133.403	95.803	56.822	51.999	110.812	104.997	144.773	179.276	118.083
RMSE	1.973	1.982	2.283	2.342	2.018	2.216	2.046	1.843	1.929
Moto: Tho table chame	marity rating	too for non o	tim protono sociento	-h immetor of	ad avnortor du	mine Ctandond	non ni) pavano	anthona) and hoto	mahadaatiaitu

parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified with data on the depth of PTAs from Dür et al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>= 4. I.e., the shallow dummy gives the effect of having an PTA, the deep dummy gives the additional effect of having a deep agreement. Note: The table shows gravity estimates for non services sectors with importer and exporter dummies. Standard errors (in parantheses) are heteroskedasticity-robust. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the coefficient on the PTA dummy is the
	(24) Water transport	-0.037	(0.024)	0.329^{***}	(0.042)	0.030^{*}	(0.018)	0.449^{***}	(0.069)	-0.042	(0.032)	0.244^{***}	(0.070)	0.323	(1.124)	0.185	(0.118)	12656	0.917	872.478	0.904
DIS	(23) Transport nec	-0.022	(0.014)	0.175^{***}	(0.027)	-0.047***	(0.013)	0.400^{***}	(0.059)	-0.073^{***}	(0.020)	0.379^{***}	(0.053)	0.713	(0.894)	0.164^{*}	(0.093)	12656	0.959	2062.090	0.556
ervice secto	(22)Trade	0.004	(0.017)	0.324^{***}	(0.031)	-0.036^{**}	(0.014)	0.409^{***}	(0.063)	-0.060**	(0.024)	0.394^{***}	(0.055)	0.902	(1.051)	0.130	(0.105)	12656	0.958	1711.045	0.646
y estimates se	(21) Construction	-0.087***	(0.027)	0.162^{***}	(0.044)	-0.087***	(0.020)	0.302^{***}	(0.081)	0.044	(0.035)	0.178^{**}	(0.070)	-0.107	(1.221)	0.069	(0.126)	12656	0.905	814.944	1.000
LS gravity	(20) Water	-0.023	(0.014)	0.181^{***}	(0.029)	-0.018	(0.013)	0.497^{***}	(0.065)	-0.037^{*}	(0.020)	0.360^{***}	(0.057)	0.689	(0.948)	0.195^{*}	(0.111)	12571	0.965	3376.432	0.560
ble A9: O	(19) Gas	0.029	(0.028)	0.195^{***}	(0.041)	0.019	(0.021)	0.507^{***}	(0.087)	0.002	(0.035)	0.321^{***}	(0.066)	1.549^{*}	(0.876)	0.423^{***}	(0.160)	9161	0.963	2052.799	0.841
Ta	(18) Electricity	-0.011	(0.022)	0.157^{***}	(0.043)	-0.087***	(0.024)	2.066^{***}	(0.158)	-0.026	(0.035)	0.386^{***}	(0.088)	0.621	(0.954)	0.187	(0.235)	11918	0.971	2825.869	0.753
		Shallow PTA		Deep PTA		Ln distance		Contiguity		Common language		Ever colonial rel.		Current colonial rel.		Ever same country		Observations	$ m R^2$	F-stat	RMSE

Note: The table shows OLS gravity estimates for services sectors with with importer and exporter fixed effects. Standard errors (in parantheses) are heteroskedasticity-robust and clustered at the country-pair level. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The
coefficient on the coefficient on the PIA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PIAs is classified with data on the
depth of PTAs from Dür et al. (2014) , which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>= 0, and the deep PTA
dummy switches to one for depth>= 4. I.e., the shallow dummy gives the effect of having an PTA, thedeep dummy gives the additional effect of having a
middle/deep agreement. Sector 32 "Dwellings" not included in the estimations because there are no trade flows reported.

	Ĩ						
	(25) Air transport	(26) Communication	(27) Financial serv. nec	(28) Insurance	(29) Business serv.	(30) Recreational serv.	(31) Other
Shallow PTA	0.006	-0.020	-0.018	-0.026^{*}	-0.031*	-0.023	-0.028*
	(0.016)	(0.012)	(0.017)	(0.014)	(0.016)	(0.015)	(0.016)
Deep PTA	0.028	0.079^{***}	0.216^{***}	0.179^{***}	0.229^{***}	0.167^{***}	0.259^{***}
	(0.027)	(0.024)	(0.033)	(0.028)	(0.029)	(0.028)	(0.029)
In distance	-0.005	-0.020	-0.075***	-0.003	-0.057 * * *	-0.044^{***}	-0.047***
	(0.013)	(0.013)	(0.015)	(0.013)	(0.013)	(0.014)	(0.014)
Contiguity	0.343^{***}	0.386^{***}	0.285^{***}	0.396^{***}	0.312^{***}	0.394^{***}	0.369^{***}
	(0.053)	(0.058)	(0.069)	(0.058)	(0.061)	(0.060)	(0.063)
Jommon language	-0.030	-0.021	-0.062^{**}	-0.034^{*}	-0.044^{**}	-0.028	-0.050^{**}
	(0.021)	(0.018)	(0.025)	(0.020)	(0.022)	(0.021)	(0.023)
Ever colonial rel.	0.217^{***}	0.237^{***}	0.271^{***}	0.181^{***}	0.273^{***}	0.270^{***}	0.287^{***}
	(0.041)	(0.050)	(0.070)	(0.055)	(0.049)	(0.052)	(0.059)
Jurrent colonial rel.	1.323	1.045	0.698	1.627^{**}	0.833	1.566^{***}	0.640
	(1.111)	(1.022)	(0.917)	(0.696)	(1.206)	(0.599)	(1.006)
Ever same country	-0.005	0.088	0.205^{*}	0.219^{**}	0.110	0.107	0.281^{***}
	(0.079)	(0.091)	(0.117)	(0.101)	(0.098)	(0.099)	(0.105)
Observations	12656	12656	12656	12656	12656	12656	12656
32	0.956	0.966	0.954	0.963	0.966	0.961	0.942
P-stat	1802.999	3271.339	1634.020	2293.654	2309.465	2412.702	1599.477
RMSE	0.611	0.488	0.680	0.555	0.604	0.591	0.618

Note: The table shows OLS gravity estimates for services sectors with with importer and exporter fixed effects. Standard errors (in parantheses) are heteroskedasticity-robust and clustered at the country-pair level. $*$, $**$ and $***$ indicate statistical significance at the 10, 5 and 1% level, respectively. The
coefficient on the coefficient on the PTA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified with data on the
depth of PTAs from Dür et al. (2014) , which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>= 0, and the deep PTA
dummy switches to one for $depth >= 4$. I.e., the shallow dummy gives the effect of having an PTA, thedeep dummy gives the additional effect of having a
middle/deep agreement. Sector 32 "Dwellings" not included in the estimations because there are no trade flows reported.

	$\begin{array}{c} (1) \\ A \sigma riculture \end{array}$	(2) Mining	(3)Food	(4)Textiles	(5) I.eather	(9) Wood	(7)	(8) Petroleum
In tomiff	2 010**	15 700***	3 071***	×**050 V	л - 20 ***	6 201 ***	A 01 2***	7 102**
	(0 608)	(2.185)	-0.211 (0.503)	-4.330 (0.611)	(0 808)	-0.351 (0 785)	(0.939)	(2,858)
PTA	0.540^{***}	(2.100) 0.244*	0.198^{**}	0.489^{***}	0.428^{***}	0.314^{***}	0.389^{***}	(000.0-
	(0.081)	(0.130)	(0.078)	(0.071)	(0.088)	(0.079)	(0.081)	(0.153)
Ln distance	-1.163^{***}	-1.670^{***}	-1.508^{***}	-1.426^{***}	-1.295^{***}	-1.624^{***}	-1.946^{***}	-2.494^{***}
	(0.051)	(0.083)	(0.045)	(0.044)	(0.054)	(0.048)	(0.051)	(0.094)
Contiguity	1.051^{***}	0.544^{***}	0.560^{***}	0.646^{***}	0.850^{***}	0.644^{***}	0.315^{*}	0.570^{**}
	(0.148)	(0.207)	(0.148)	(0.155)	(0.180)	(0.158)	(0.166)	(0.242)
Common language	0.514^{***}	0.174	0.676^{***}	0.710^{***}	0.666^{***}	0.567^{***}	1.135^{***}	-0.237
	(0.095)	(0.151)	(060.0)	(0.083)	(0.101)	(0.089)	(0.098)	(0.184)
Ever colonial rel.	1.091^{***}	1.144^{***}	0.974^{***}	0.748^{***}	0.880^{***}	0.946^{***}	0.837^{***}	0.730^{***}
	(0.140)	(0.204)	(0.141)	(0.142)	(0.156)	(0.132)	(0.140)	(0.265)
Current colonial rel.	0.074	1.571^{**}	-1.833	0.209	0.581	-0.294	-3.075	2.166
	(0.322)	(0.675)	(2.386)	(2.411)	(2.540)	(2.495)	(2.452)	(1.385)
Ever same country	0.208	0.543	0.693^{***}	0.406	0.758^{***}	0.310	0.487^{**}	0.505
	(0.222)	(0.360)	(0.214)	(0.252)	(0.288)	(0.233)	(0.228)	(0.388)
Observations	8132	5384	8225	8766	6948	7534	7874	4191
${ m R}^2$	0.620	0.540	0.670	0.756	0.699	0.717	0.736	0.529
F-stat	65.028	36.062	77.099	120.710	81.338	94.352	111.040	
RMSE	2.233	2.943	2.082	1.946	2.137	2.032	2.156	3.052

Note: The table shows OLS gravity estimates for non services sectors with importer and exporter dumnies.	Standard errors (in parantheses) are
heteroskedasticity-robust. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively.	The coefficient on the PTA dummy is
the parameter estimate for δ/θ .	

	Table A12:	OLS gravit	ty estimates ma	anufacturing	g sectors wit]	h simple PTA	dummy, cor	at'd.	
	(9) Chemicals	(10) Minerals	(11) Ferrous metals	(12) Metal nec	(13) Metal prod.	(14) Transp. equip	(15) Electronics	(16) Machinery nec	(17) Manufactures nec
In tariff	-10 578***	-8 007***	-0 300***	-19 6/5***	- 4 180***	-0 130***	-9 A50**	-6 680***	-3 707***
	(0.855)	(0.856)	(1.265)	(1.521)	(0.923)	(0.725)	(1.164)	(0.855)	(0.728)
PTA	0.499^{***}	0.285^{***}	0.641^{***}	0.368^{***}	0.364^{***}	0.832^{***}	0.361^{***}	0.505^{***}	0.276^{***}
	(0.063)	(0.080)	(0.095)	(0.103)	(0.076)	(0.084)	(0.077)	(0.064)	(0.072)
Ln distance	-1.544^{***}	-1.662^{***}	-1.624^{***}	-1.709^{***}	-1.693^{***}	-1.352^{***}	-1.262^{***}	-1.409^{***}	-1.309^{***}
	(0.041)	(0.049)	(0.057)	(0.064)	(0.046)	(0.052)	(0.047)	(0.040)	(0.045)
Contiguity	0.635^{***}	0.943^{***}	0.667^{***}	0.162	0.379^{**}	0.738^{***}	0.292^{*}	0.533 * * *	0.407^{***}
	(0.151)	(0.165)	(0.159)	(0.181)	(0.154)	(0.155)	(0.174)	(0.155)	(0.153)
Common language	0.765^{***}	0.708^{***}	0.341^{***}	0.560^{***}	0.831^{***}	0.645^{***}	0.558^{***}	0.728^{***}	0.818^{***}
	(0.080)	(0.093)	(0.115)	(0.119)	(060.0)	(0.099)	(0.089)	(0.075)	(0.086)
Ever colonial rel.	0.412^{***}	0.777^{***}	0.579^{***}	1.201^{***}	0.888^{***}	0.839^{***}	0.976^{***}	0.812^{***}	0.893^{***}
	(0.113)	(0.139)	(0.155)	(0.163)	(0.130)	(0.143)	(0.144)	(0.123)	(0.144)
Current colonial rel.	-0.329	-2.106	0.976	1.331	-0.934	-0.074	-0.521	0.287	0.488
	(2.389)	(3.728)	(2.104)	(1.028)	(2.373)	(0.986)	(3.507)	(2.463)	(2.241)
Ever same country	0.398*	0.469^{*}	0.518^{**}	-0.351	0.437*	0.189	0.248	0.372	0.523^{**}
	(0.224)	(0.259)	(0.258)	(0.282)	(0.238)	(0.243)	(0.287)	(0.227)	(0.221)
Observations	9248	7278	6001	5682	7943	7510	8162	9316	8025
${ m R}^2$	0.767	0.731	0.666	0.652	0.747	0.728	0.770	0.809	0.744
F-stat	135.617	96.128	56.864	51.757	112.306	107.944	144.913	182.043	118.948
RMSE	1.971	1.981	2.283	2.345	2.018	2.208	2.047	1.838	1.932
Note: The table show	vs OLS gravi	ty estimates	s for non service	s sectors wit	h importer an	d exporter dum	mies. Stands	ard errors (in pa	trantheses) are
Hereroskedasuichty-ront	nsr. , , and	a marca	the statistical sign	uncance at tr	le lu, o and l	70 level, respect	Ivery. LITE COC	ellicient on the r	TA duffing is

Note: The table shows OLS gravity estimates for non services sectors with importer an	d exporter dummies.	Standard er
heteroskedasticity-robust. *, ** and *** indicate statistical significance at the 10, 5 and 1	% level, respectively.	The coefficien

TAUL	CID : OLD	gravity ex	sumates se	SUVICES SECTORS	MINI SIIII	ne r ta aumi	ıy
	(18)	(19)	(20)	(21)	(22)	(23)	(24)
	Electricity	$_{\rm Gas}$	Water	Construction	Trade	Transport nec	Water transport
PTA	0.018	0.070^{***}	0.009	-0.058^{**}	0.061^{***}	0.009	0.021
	(0.021)	(0.026)	(0.014)	(0.026)	(0.016)	(0.014)	(0.023)
Ln distance	-0.110^{***}	-0.009	-0.045^{***}	-0.111^{***}	-0.083***	-0.072***	-0.018
	(0.023)	(0.020)	(0.013)	(0.019)	(0.013)	(0.012)	(0.017)
Contiguity	2.046^{***}	0.484^{***}	0.475^{***}	0.281^{***}	0.368^{***}	0.378^{***}	0.407^{***}
	(0.159)	(0.088)	(0.065)	(0.082)	(0.064)	(0.059)	(0.069)
Common language	-0.033	-0.008	-0.045^{**}	0.037	-0.074^{***}	-0.080***	-0.056^{*}
	(0.035)	(0.035)	(0.020)	(0.035)	(0.024)	(0.020)	(0.032)
Ever colonial rel.	0.372^{***}	0.306^{***}	0.345^{***}	0.165^{**}	0.367^{***}	0.365^{***}	0.217^{***}
	(0.088)	(0.066)	(0.058)	(0.071)	(0.055)	(0.053)	(0.071)
Current colonial rel.	0.619	1.565^{*}	0.693	-0.104	0.907	0.717	0.329
	(0.981)	(0.911)	(0.980)	(1.250)	(1.113)	(0.924)	(1.184)
Ever same country	0.196	0.423^{***}	0.205^{*}	0.080	0.152'	0.175^{*}	0.207*
	(0.234)	(0.161)	(0.111)	(0.126)	(0.104)	(0.093)	(0.118)
Observations	11918	9161	12571	12656	12656	12656	12656
R^2	0.971	0.963	0.964	0.905	0.958	0.959	0.917
F-stat	2836.325	2056.225	3424.509	820.488	1727.853	2078.685	875.555
RMSE	0.754	0.842	0.562	1.001	0.649	0.557	0.906

are	the the	
heses)	imy is	
parant	A dun	
i (in 1	le PT.	
errors	on th	
ndard	ficient	ed.
. Sta	le coef	report
effects.	y. Th	flows
ixed e	ectivel	trade
orter f	, resp	re no
l expc	level	nere a
er anc	nd 1%	ause tl
nport	0, 5 a.	s bec
vith iı	the 1	nation
with v	nce at	e estir
ctors 7	nificar	in th
es sec	al sig	sluded
servic	atistic	tot inc
ss for	ate st	ngs" n
timate	indic	Dwellin
ity est	*** pi	.32 "I
grav	** an	Sector
s OLS	st. *,	δ/θ .
show	/-robu	te for
table	sticity	estima
The	skeda	leter €
Note:	heterc	paran

	(25) Air transport	(26) Communication	(27) Financial serv. nec	(28) Insurance	(29) Business serv.	(30) Recreational serv.	(31) Other
PTA	0.011	-0.006	0.020	0.006	0.010	0.007	0.017
	(0.015)	(0.012)	(0.017)	(0.014)	(0.016)	(0.015)	(0.016)
Ln distance	-0.009	-0.032^{***}	-0.106^{***}	-0.029^{**}	-0.090^{***}	-0.068***	-0.085***
	(0.012)	(0.012)	(0.015)	(0.012)	(0.012)	(0.013)	(0.013)
Contiguity	0.340^{***}	0.376^{***}	0.258^{***}	0.374^{***}	0.283^{***}	0.373^{***}	0.336^{***}
	(0.053)	(0.058)	(0.069)	(0.058)	(0.061)	(0.060)	(0.064)
Common language	-0.031	-0.025	-0.071^{***}	-0.041^{**}	-0.053^{**}	-0.035*	-0.060***
	(0.021)	(0.018)	(0.025)	(0.020)	(0.022)	(0.021)	(0.023)
Ever colonial rel.	0.214^{***}	0.231^{***}	0.254^{***}	0.166^{***}	0.254^{***}	0.257^{***}	0.265^{***}
	(0.041)	(0.050)	(0.070)	(0.056)	(0.049)	(0.052)	(0.059)
Current colonial rel.	1.324	1.046	0.702	1.630^{**}	0.837	1.569^{**}	0.644
	(1.117)	(1.037)	(0.958)	(0.730)	(1.250)	(0.628)	(1.052)
Ever same country	-0.003	0.093	0.219^{*}	0.231^{**}	0.125	0.118	0.298^{***}
	(0.079)	(0.091)	(0.117)	(0.101)	(0.097)	(0.099)	(0.105)
Observations	12656	12656	12656	12656	12656	12656	12656
$ m R^2$	0.956	0.966	0.954	0.963	0.965	0.961	0.941
F-stat	1811.438	3298.179	1644.272	2285.526	2327.854	2427.491	1616.104
RMSE	0.611	0.488	0.681	0.556	0.606	0.592	0.620

are	the	
es)	.is	
hes	umy	
rant	dun	
pa	ΓA	
(in	е Д	
rors	ı th	
l er	t oi	
larc	cien	ч.
tan	oeffi	orte
S.	le ci	rep(
ects	Ē	SMO
effe	ely.	e flc
ixed	ctiv	rad
er fi	espe	no t
ort	el, r	are
exl	leve	ere
and	1%	e th
ter	and	caus
lpor	ъ.	bec
i im	e 10	ions
with	the	mat
ith	se at	esti
N N	canc	the
ctor	nifi	l in
s se	l sig	ndec
vice	tica	incl
ser	atis	lot
for	se st	
ates	licat	lling
tim	.ŭ	Dwe
y es	* * *	2 "]
avit	and	or 3
10 10	* *	Sect
OLS	*`	.θ
MS ()ust.) r $\delta_{/}$
sho	-rot	te fc
ble	icity	ima
e ta	lasti	est
Th	skec	eter
te:	tero:	ram
$^{\rm N}_{ m o}$	het	pa

ALCONT.		~						
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	Agriculture	Mining	Food	Textiles	Leather	Wood	Paper	$\operatorname{Petroleum}$
n tariff	-2.779^{***}	-13.071^{***}	-2.392^{***}	-2.158^{***}	-3.236^{***}	-4.336^{***}	-4.664^{***}	-3.991
	(0.620)	(2.241)	(0.526)	(0.728)	(0.856)	(0.832)	(0.925)	(2.918)
TA	1.942^{***}	2.213^{***}	1.090^{***}	1.612^{***}	1.445^{***}	1.625^{***}	0.815^{***}	1.239^{***}
	(0.224)	(0.354)	(0.192)	(0.195)	(0.214)	(0.194)	(0.183)	(0.336)
n distance	-0.731^{***}	-1.066^{***}	-1.239^{***}	-1.103^{***}	-1.000^{***}	-1.218***	-1.812***	-2.116^{***}
	(0.085)	(0.132)	(0.071)	(0.071)	(0.081)	(0.074)	(0.075)	(0.133)
ontiguity	1.035^{***}	0.540^{***}	0.555^{***}	0.645^{***}	0.866^{***}	0.688^{***}	0.317^{*}	0.591^{**}
	(0.150)	(0.210)	(0.145)	(0.156)	(0.180)	(0.161)	(0.164)	(0.239)
ommon language	0.534^{***}	0.117	0.698^{***}	0.742^{***}	0.696^{***}	0.601^{***}	1.147^{***}	-0.261
	(0.096)	(0.152)	(0.088)	(0.083)	(0.100)	(0.090)	(0.097)	(0.180)
ver colonial rel.	1.188^{***}	1.310^{***}	1.030^{***}	0.821^{***}	0.956^{***}	1.029^{***}	0.867^{***}	0.833^{***}
	(0.144)	(0.204)	(0.138)	(0.138)	(0.153)	(0.133)	(0.138)	(0.259)
urrent colonial rel.	-0.128	1.519^{**}	-1.958	-0.028	0.416	-0.510	-3.137	2.046
	(0.291)	(0.610)	(2.339)	(2.340)	(2.477)	(2.419)	(2.397)	(1.285)
ver same country	0.214	0.587^{*}	0.693^{***}	0.440^{*}	0.793^{***}	0.342	0.485^{**}	0.573
	(0.226)	(0.348)	(0.205)	(0.246)	(0.288)	(0.234)	(0.222)	(0.380)
bservations	8132	5384	8225	8766	6948	7534	7874	4191
5	0.605	0.519	0.664	0.748	0.692	0.705	0.735	0.521
-stat	64.453	34.436	75.215	116.889	80.085	90.621	109.497	97.838
MSE	2.245	2.945	2.071	1.950	2.124	2.041	2.128	2.991
-stat weak-ID	803.582	605.267	962.131	910.031	834.970	852.338	949.115	684.502
under-ID	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000

Note: The table shows IV gravity estimates for non services sectors with importer and exporter dummies. Standard errors (in parantheses) are heteroskedasticity-robust. PTA dummy instrumented with a contagion index a la Martin et al. (2012) or Baldwin and Jaimovich (2012). *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the PTA dummy is the parameter estimate for δ/θ .

	Table A16:	IV gravity	v estimates mai	nutacturing	sectors with	i simple P'I'A c	dummy, cont	t'd.	
	(6)	(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)
	Chemicals	Minerals	Ferrous metals	Metal nec	Metal prod.	Transp. equip	Electronics	Machinery nec	Manufactures nec
Ln tariff	-10.161^{***}	-8.458***	-8.467^{***}	-10.856^{***}	-4.683^{***}	-1.938^{***}	-2.357^{**}	-6.543^{***}	-3.626^{***}
	(0.855)	(0.866)	(1.253)	(1.625)	(0.938)	(0.719)	(1.153)	(0.851)	(0.739)
PTA	1.023^{***}	0.665^{***}	1.426^{***}	1.036^{***}	0.701^{***}	1.174^{***}	0.646^{***}	0.741^{***}	0.425^{**}
	(0.169)	(0.188)	(0.198)	(0.250)	(0.174)	(0.195)	(0.185)	(0.163)	(0.175)
Ln distance	-1.380^{***}	-1.543^{***}	-1.386^{***}	-1.514^{***}	-1.589^{***}	-1.243^{***}	-1.172^{***}	-1.335^{***}	-1.263^{***}
	(0.067)	(0.075)	(0.070)	(0.092)	(0.068)	(0.077)	(0.073)	(0.065)	(0.069)
Contiguity	0.629^{***}	0.947^{***}	0.677^{***}	0.169	0.382^{**}	0.739^{***}	0.289^{*}	0.532^{***}	0.408^{***}
	(0.149)	(0.162)	(0.158)	(0.180)	(0.151)	(0.152)	(0.171)	(0.153)	(0.150)
Common language	0.778^{***}	0.717^{***}	0.349^{***}	0.549^{***}	0.841^{***}	0.653^{***}	0.562^{***}	0.736^{***}	0.822^{***}
	(0.079)	(0.092)	(0.114)	(0.116)	(0.088)	(0.097)	(0.088)	(0.075)	(0.085)
Ever colonial rel.	0.446^{***}	0.802^{***}	0.638^{***}	1.256^{***}	0.908^{***}	0.860^{***}	0.995^{***}	0.826^{***}	0.901^{***}
	(0.113)	(0.135)	(0.156)	(0.161)	(0.127)	(0.139)	(0.141)	(0.121)	(0.142)
Current colonial rel.	-0.428	-2.166	0.938	1.311	-0.992	-0.088	-0.545	0.245	0.461
	(2.332)	(3.658)	(2.045)	(0.980)	(2.328)	(0.957)	(3.443)	(2.421)	(2.203)
Ever same country	0.394^{*}	0.482^{*}	0.535^{**}	-0.338	0.442^{*}	0.190	0.247	0.370^{*}	0.528^{**}
	(0.218)	(0.253)	(0.252)	(0.278)	(0.233)	(0.238)	(0.284)	(0.223)	(0.217)
Observations	9248	7278	6001	5682	7943	7510	8162	9316	8025
$ m R^2$	0.765	0.730	0.662	0.649	0.746	0.727	0.769	0.809	0.743
F-stat	133.998	95.594	56.661	51.660	110.146	107.787	144.314	179.306	118.439
RMSE	1.953	1.953	2.253	2.306	1.991	2.176	2.019	1.817	1.904
F-stat weak-ID	1079.388	878.839	868.955	698.166	944.051	913.989	914.374	1025.900	917.442
p under-ID	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000
Moto: The table che	IV anonity	. octimotod		contour with	immonton ond	and an during	mine Ctondo	or uit presented	manthoron) and

Note: The table shows IV gravity estimates for non services sectors with importer and exporter dummies. Standard errors (in parantheses) are heteroskedasticity-robust. PTA dummy instrumented with a contagion index a la Martin et al. (2012) or Baldwin and Jaimovich (2012). *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the PTA dummy is the parameter estimate for δ/θ .

	(18)	(19)	(20)	(21)	(22)	(23)	(24)
	Electricity	Gas	Water	Construction	Trade	Transport nec	Water transport
PTA	0.169^{**}	0.122	0.112^{**}	-0.036	0.314^{***}	0.158^{***}	0.110
	(0.072)	(0.076)	(0.047)	(0.082)	(0.056)	(0.046)	(0.077)
Ln distance	-0.064^{*}	0.008	-0.013	-0.106^{***}	-0.005	-0.027	0.010
	(0.033)	(0.031)	(0.019)	(0.030)	(0.021)	(0.018)	(0.028)
Contiguity	2.039^{***}	0.483^{***}	0.470^{***}	0.283^{***}	0.354^{***}	0.371^{***}	0.402^{***}
	(0.158)	(0.087)	(0.065)	(0.081)	(0.064)	(0.059)	(0.069)
Common language	-0.029	-0.006	-0.041^{**}	0.034	-0.064^{***}	-0.076^{***}	-0.052*
	(0.035)	(0.035)	(0.020)	(0.034)	(0.024)	(0.020)	(0.032)
Ever colonial rel.	0.383^{***}	0.314^{***}	0.352^{***}	0.163^{**}	0.387^{***}	0.375^{***}	0.224^{***}
	(0.087)	(0.066)	(0.058)	(0.070)	(0.057)	(0.054)	(0.071)
Current colonial rel.	0.585	1.544^{*}	0.665	-0.095	0.830	0.678	0.301
	(0.966)	(0.894)	(0.967)	(1.240)	(1.090)	(0.910)	(1.169)
Ever same country	0.193	0.419^{***}	0.203^{*}	0.082	0.142	0.170^{*}	0.203^{*}
	(0.232)	(0.159)	(0.110)	(0.125)	(0.104)	(0.093)	(0.117)
Observations	11918	9161	12571	12656	12656	12656	12656
${ m R}^2$	0.971	0.963	0.964	0.905	0.957	0.958	0.917
RMSE	0.750	0.833	0.559	0.991	0.648	0.556	0.898

are	* * *	32	
es)	nd	ctor	
thes	യ * *	s_{e}	
uran.	*`	δ/θ .	
ı pê	$(\overline{2})$.	for	
s (ii	(201)	ate	
rror	ich	stim	
d e	mov	er er	
ndaı	Jai	mete	
Sta	and	araı	
ts.	win	he t	
effec	3ald	is t	
eq	or I	ymt	
, fix	12)	dun	
ortei	(20)	ΤĀ	
expe	al.	le P	
pu	n et	n th	
er a	Iarti	nt o	
port	la N	ficie	rted
im	k K	coef	epo
with	nde	Γhe	I SWO
ith	ion j	Ň	le flo
S. W	ltag	tive.	trac
ecto	l COI	spec	e no
es S	ith ε	l, re	e ar(
rvice	d W.	leve.	there
r se	ente	1%	use 1
s fo	um.	nd	eca
nate	instı	ы го	ns b
estir	my	10	latio
ity	lum	the	$_{ m stim}$
gravi	N N	e at	he e
N	ΓЧ	canc	in t]
ws]	ust.	snific	ded
$_{\rm sho}$	-rob	l sig	nclu
uble	icity	stica	ot iı
le tê	lasti	tatis	s" n
$^{\mathrm{Th}}$	skec	te s	lling
ote:	sterc	dica	Dwel
ž	he	ΞŪ.	Ì,

)					
	(25)	(26)	(27)	(28)	(29)	(30)	(31)
	Air transport	Communication	Financial serv. nec	Insurance	Business serv.	Recreational serv.	Other
PTA	-0.062	-0.029	0.226^{***}	0.093^{**}	0.271^{***}	0.117^{**}	0.263^{***}
	(0.050)	(0.043)	(0.057)	(0.045)	(0.052)	(0.048)	(0.053)
Ln distance	-0.030	-0.039^{**}	-0.043^{**}	-0.002	-0.011	-0.035*	-0.011
	(0.019)	(0.018)	(0.022)	(0.018)	(0.020)	(0.019)	(0.021)
Contiguity	0.342^{***}	0.377^{***}	0.248^{***}	0.370^{***}	0.271^{***}	0.368^{***}	0.325^{***}
	(0.052)	(0.057)	(0.069)	(0.058)	(0.062)	(0.060)	(0.064)
Common language	-0.032	-0.026	-0.065^{***}	-0.039*	-0.047^{**}	-0.032	-0.054^{**}
	(0.021)	(0.018)	(0.025)	(0.020)	(0.022)	(0.021)	(0.023)
Ever colonial rel.	0.210^{***}	0.229^{***}	0.268^{***}	0.172^{***}	0.271^{***}	0.264^{***}	0.282^{***}
	(0.040)	(0.049)	(0.070)	(0.056)	(0.051)	(0.053)	(0.061)
Current colonial rel.	1.339	1.053	0.646	1.608^{**}	0.771	1.540^{**}	0.580
	(1.109)	(1.028)	(0.940)	(0.720)	(1.228)	(0.618)	(1.033)
Ever same country	-0.002	0.094	0.212^{*}	0.228^{**}	0.117	0.115	0.290^{***}
	(0.078)	(0.090)	(0.116)	(0.100)	(0.099)	(0.090)	(0.104)
Observations	12656	12656	12656	12656	12656	12656	12656
\mathbb{R}^2	0.956	0.966	0.953	0.963	0.965	0.961	0.940
RMSE	0.605	0.484	0.677	0.552	0.606	0.588	0.619

are	*	32	
(Se	* pu	ctor	
chese	ਲ *	\tilde{s}	
rant	*`	δ/θ .	
ı pa	2).	for	
	(201)	ate i	
rrors	ich	tim	
id e	mov	er es	
ndaı	Jai	mete	
Sta	and	araı	
ts.	win	he t	
effec	3ald	is t	
eq	or I	amy	
r fix	12)	dun	
orte	(20)	TA	
\exp	t al.	he F	
and	in e	on t	
ter	Mart	ent	÷
ıpor	la l	fficie	ortec
h in	ex a	coe	repo
wit]	inde	$_{\rm The}$	SWO
vith	gion	ely.	de fl
ors v	ontag	ctive	o tra
secto	a cc	espe	re no
ces :	vith	el, r	re ai
ervi	ed v	lev	the
ors	nent	11%	ause
les f	trun	and	\mathbf{bec}
imat	r ins	0, 5	ions
est	nmy	he 1	mat
wity	mp .	at t	esti
gra	PTA	nce	the
S IV	št.]	ifica	in in
how	;nqo	sign	slude
le s	ity-r	ical	t inc
tab	astic.	atist.	, no
The	kedε	$e st_{\delta}$	ings'
te:	eros	icat	welli
Noi	het	ind	Ĝ

		Non-servi	ce sectors			Service	sectors	
	I			LS	Ц		IO	S
		weighted		weighted		weighted		weighted
Ln tariff	-5.630^{***}	-5.606***	-5.665^{***}	-5.546^{**}				
	(0.195)	(0.237)	(0.183)	(0.212)				
Shallow PTA	0.438^{***}	0.364^{*}	0.167^{***}	0.204^{***}	0.076^{*}	0.053^{*}	-0.036^{*}	-0.023*
	(0.166)	(0.212)	(0.025)	(0.031)	(0.046)	(0.031)	(0.021)	(0.014)
Deep PTA	0.999^{***}	0.994^{***}	0.426^{***}	0.385^{***}	0.220^{***}	0.273^{***}	0.202^{***}	0.211^{***}
	(0.221)	(0.283)	(0.033)	(0.042)	(0.070)	(0.046)	(0.034)	(0.023)
Ln distance	-1.163^{***}	-1.150^{***}	-1.356^{***}	-1.320^{***}	-0.008	-0.013	-0.041^{***}	-0.045^{***}
	(0.023)	(0.029)	(0.015)	(0.019)	(0.020)	(0.013)	(0.015)	(0.010)
Contiguity	0.878^{***}	0.893^{***}	0.774^{***}	0.787^{***}	0.503^{***}	0.389^{***}	0.500^{***}	0.381^{***}
	(0.049)	(0.063)	(0.045)	(0.057)	(0.053)	(0.040)	(0.052)	(0.040)
Common language	0.564^{***}	0.564^{***}	0.545^{***}	0.541^{***}	-0.040	-0.049^{**}	-0.037	-0.048^{**}
	(0.031)	(0.038)	(0.030)	(0.036)	(0.028)	(0.019)	(0.028)	(0.019)
Ever colonial rel.	0.837^{***}	0.807^{***}	0.801^{***}	0.769^{***}	0.255^{***}	0.291^{***}	0.254^{***}	0.285^{***}
	(0.052)	(0.064)	(0.048)	(0.058)	(0.062)	(0.043)	(0.061)	(0.043)
Current colonial rel.	0.128	0.230	0.135	0.235	0.763	0.862^{*}	0.761	0.861^{*}
	(0.527)	(0.660)	(0.546)	(0.689)	(0.549)	(0.506)	(0.550)	(0.510)
Ever same country	0.273^{***}	0.225^{**}	0.243^{***}	0.213^{**}	0.212^{***}	0.136^{**}	0.211^{***}	0.139^{**}
	(0.070)	(0.092)	(0.067)	(0.087)	(0.076)	(0.063)	(0.076)	(0.062)
Observations	122971	122971	126219	126219	172866	172866	172866	172866
$ m R^2$	0.518	0.546	0.519	0.546	0.442	0.736	0.442	0.736
F-stat			709.131	490.729			749.754	1723.141
RMSE	2.802	2.769	2.795	2.766	2.853	1.680	2.855	1.681

aweights, where the weight is the share of a sector in world trade. Non-service sectors in GTAP sample, 0.5% of the observations with the highest tariffs Jaimovich (2012), historical and recent war frequency and lagged average variables for political similarity (average of 2000-2005). Column (2) and (4) use in each sector dropped from sample. *, ** and *** indicate statistical significance at the 10, 5 and 1% level, respectively. The coefficient on the coefficient al. (2014), which ranges from 0 to 7 and where the shallow PTA dummy switches to one for depth>= 0, and the deep PTA dummy switches to one for are heteroskedasticity-robust. In columns (1) and (2), PTA dummies instrumented with single contagion index a la Martin et al. (2012) or Baldwin and on the PTA dummy is the parameter estimate for δ/θ . The dummy for shallow and deep PTAs is classified with data on the depth of PTAs from Dür et Note: The table shows gravity estimates for non services sectors on pooled sectoral data with importer and exporter dummies. Standard errors (in parantheses) depth>= 4. I.e., the shallow dummy gives the effect of having an PTA, the deep dummy gives the additional effect of having a deep agreement.

Table A19: Pooled regressions

C Sensitivity checks: welfare effects from alternative empirical specifications

The results presented so far rely on the PTA coefficients and θ s estimated with an IV approach and using two different depths of PTAs. Next, we investigate the sensitivity of the results pertaining to this choice. Table A20 presents results for three different empirical specifications. In column (1) and (2), the simulations are based on a single PTA dummy. In Column (1), the PTA dummy is instrumented with the contagion index. Column (2) is based on OLS regression. By and large, the predicted region-specific growth rates of real income are stable. The correlation of the country-specific results in the baseline with the instrumented and uninstrumented results are very high, with a correlation coefficient of 0.97 in both cases, statistically significant at the 1% level. With a single PTA dummy, the expected effects are slightly larger. But the overall tendencies are similar, except for ASEAN and China. For these two regions, the baseline predicts negative results of around 0.2%, while the effects are positive and in the order of magnitude of 0.1% with a single PTA dummy.

The real income changes in column (4) are based on trade elasticity estimates including a shallow and deep PTA dummy, but the PTA dummies are not instrumented. The correlation coefficient of the country-specific results in the baseline with this specification is again very high, 0.98 and statistically highly significant. The sorting of the regions is very similar, with the largest negative effects in China and the ASEAN region. But the predicted effects are substantially smaller. TTIP is still predicted to be positive for TTIP countries, on average, but it only raises real income by 0.4%.

Last, column (5) presents results when sectoral heterogeneity in the treatment intensity is supressed. We run two estimations on pooled sectoral data, one for the manu-

	(1)	(2)	(3)	(4)	(5)
		V	Velfare growth	(in %)	
Estimation:	Base	Single	PTA dummy	Shal	\log/Deep
	IV	IV	OLS	OLS	Pooled IV
TTIP	2.37	2.46	2.80	0.42	1.15
USA	2.68	2.80	3.17	0.47	1.27
EU27	2.12	2.17	2.50	0.38	1.04
Non-TTIP	-0.03	0.06	0.08	-0.03	-0.13
Rest of World	0.40	0.51	0.60	0.06	0.18
Rest of Europe	0.29	0.40	0.48	0.03	0.07
Central Asia	0.20	0.29	0.34	0.02	0.02
Eurasian Customs Union	0.17	0.19	0.23	0.02	0.02
$SACU^{a}$	0.15	0.20	0.24	-0.00	-0.05
EFTA^{b}	0.15	0.26	0.31	-0.02	-0.13
Canada	0.12	0.13	0.17	-0.02	-0.17
$MENA^{c}$	0.12	0.20	0.24	-0.00	-0.03
Latin America & Caribbean	0.11	0.15	0.17	0.02	0.06
Sub-Saharan Africa	0.10	0.19	0.22	-0.00	-0.03
Turkey	0.10	0.12	0.15	-0.00	-0.01
Oil exporters	0.09	0.26	0.30	-0.04	-0.19
Australia & New Zealand	0.08	0.12	0.14	-0.00	-0.02
Oceania	0.06	0.14	0.16	-0.01	-0.04
South Asia	0.06	0.13	0.15	0.00	0.01
MERCOSUR	0.00	0.03	0.04	-0.02	-0.05
Pacific Alliance	-0.07	-0.07	-0.06	-0.06	-0.25
East Asia	-0.13	-0.14	-0.16	-0.04	-0.14
$ASEAN^d$	-0.19	0.06	0.07	-0.07	-0.29
China	-0.23	0.08	0.10	-0.08	-0.29
World	1.32	1.40	1.60	0.22	0.58

Table A20: Sensitivity check: Results with alternative empirical specifications

Note: The regional real income change is a GDP-weighted sum of the respective countries' real income changes. An overview of the classification of countries into regions is provided in the online appendix. Base: deep TTIP scenario. Columns with IV based on instrumental variables regression. Columns with OLS based on ordinary least squares regression. Column (5) based on pooled sectoral estimation, i.e. all manufacturing sectors have the same estimated productivity dispersion and PTA treatment, the same is true for all service sectors.

 d Association of Southeast Asian Nations.

facturing sector, one for the service industries.². The average productivity dispersion is

 $^{^2\}mathrm{The}$ regression results are given in Table A18 in the online appendix

approximately 1/5.6. This finding is in line with the literature, which typically finds estimates of the elasticity of substitution of between 5-7, (see e.g. Broda and Weinstein, 2006; Bergstrand et al., 2012). Shallow treaties have positive effects on bilateral trade, but the empirical evidence is less strong for service industries. Deep agreements, on the other hand, have a strong positive and statistically significant effect in both manufacturing and services. The estimates suggest, that a deep PTA increases bilateral manufacturing trade by 1.7 or 170% ($\equiv e^{0.999} - 1$) and service trade by around 25%. On the country level, the correlation of real income changes with the base scenario is again very high, 0.92, and statistically significant. For TTIP countries, the effects are in the order of magnitude of 1% real income increase. So they are smaller than in the baseline. The negative effect on non-TTIP countries is more pronounced, with -0.13% on average. China and the ASEAN region are again the biggest losers.

References

- Baldwin, Richard and Dany Jaimovich, "Are Free Trade Agreements contagious?," Journal of International Economics, 2012, 88 (1), 1 – 16.
- Bergstrand, Jeffrey H., Peter Egger, and Mario Larch, "Gravity Redux: Estimation of gravity-equation coefficients, elasticities of substitution, and general equilibrium comparative statics under asymmetric bilateral trade costs," *Journal of International Economics*, 2012.
- Broda, C. and D. Weinstein, "Globalization and the Gains from Variety," *Quarterly Jour*nal of Economics, May 2006, 121 (2), 541–585.
- Dür, Andreas, Leonardo Baccini, and Manfred Elsig, "The Design of International Trade Agreements: Introducing a New Database," *Review of International Organizations*, 2014, 9 (3), 353–375.
- Martin, Philippe, Thierry Mayer, and Mathias Thoenig, "The Geography of Conflicts and Regional Trade Agreements," American Economic Journal: Macroeconomics, October 2012, 4 (4), 1–35.